SBC Long Distance, LLC
d/b/a AT&T Long Distance
NY PSC Tariff No. 1 – Telephone

#### PRICE LIST

#### 1.1 Operator Toll Assistance Services

- 1.1.1 Access Method Toll Free Access Number
  - (A) Reserved for future use
  - (B) Group 2 Toll Free Access Numbers
    - .1 Billed To LEC Card

If charges are billed to a LEC Card, the usage rates and per call charges are the same as the usage rates and per call charges described in Section 1.1.2 (A), (B), and (C) of this Price List.

- .2 Billed to Calling Card
  - .a Per Call Charges

For per call charges, see Section 1.1.2 (C) of this Price List.

- For fully automated calls billed to the Calling Card Option 3, a per call charge does not apply.
- The per call charges for all other calls billed to the Calling Card are located in Section 1.1.2 (B) and 1.1.2 (C) of this Price List.
- .b Fully Automated Usage Charges
  - i Calling Card Option 1, Option 2, and Option 4, Value Card Plus

The usage rate may be found in Section 1.1.2 (A) of this Price List.

.ii Calling Card - Option 2 Categories

Calling card usage rates for Customers that subscribe to Calling Card - Option 2 categories are the same as the usage rates that apply to 1+ outbound calls originating via Switched Access and billed under the optional calling plan selected by the Customer. Usage rates may be found in Section 1 of this Price List.

SBC Long Distance, LLC d/b/a AT&T Long Distance NY PSC Tariff No. 1 – Telephone Attachment I Original Page 2

### PRICE LIST

- 1.1 Operator Toll Assistance Services (continued)
  - 1.1.1 Access Method Toll Free Access Number (continued)
    - (B) Group 2 Toll Free Access Numbers (continued)
      - .2 Billed to Calling Card (continued)
        - .b Fully Automated Usage Charges (continued)
          - .iii Calling Card Option 3, and Option 3 Categories
            - (1) Calling card usage rates for Customers that subscribe to any of the Company's High Volume Calling plans are the same as the usage rates that apply to 1+ outbound calls originating via Switched Access and billed under the High Volume Calling optional calling plan selected by the Customer. Usage rates may be found in Section 1 of the Price List.
            - (2) Calling Card Rates for Customers That Subscribe to High Volume Dedicated Outbound Calling or High Volume Dedicated Toll Free Calling

The per minute usage rates are the same as Section 1.1.1 (B).2.b.iii (1) of the Price List.

(3) Calling Card Rates for Customers That Subscribe To High Volume Outbound Calling II or High Volume Toll Free Calling II

The per minute usage rates are the same as Section 1.7.2 (A) of the Price List.

(4) Calling Card Rates for Customers That Subscribe to High Volume Dedicated Outbound Calling II or High Volume Dedicated Toll Free Calling II

The per minute usage rates are the same as Section 1.7.2 (A) of the Price List.

.3 Billed To All Other Operator Toll Assistance Billing Options

For all other Operator Toll Assistance Services calls completed via a Group 2 Toll Free Access Number, the usage rates and per call charges are the same as the usage rates and per call charges described in Section

Effective: March 12, 2010

SBC Long Distance, LLC	Attachment I
d/b/a AT&T Long Distance	2 <sup>nd</sup> Revised Page 3
NY PSC Tariff No. 1 – Telephone	Superseding 1 <sup>st</sup> Revised Page 3

## PRICE LIST

- 1.1 Operator Toll Assistance Services (continued)
  - 1.1.2 All Other Access Methods

There are two rate elements. They include a usage charge and a per call charge. The usage charges and per call charges follow:

(A) Usage Rates

Call Type	Rate Per Minute		
LEC Card	\$0.35		
Calling Card - Option 1	\$0.90		
Calling Card - Option 2	\$0.67(T)		
Rate Change Effective April 12, 2010 (T)	\$0.70(I)		
Calling Card - Option 3	See Section 1.1.1 (B).2.b.iii		
Calling Card - Option 4	\$0.15		
All Other Operator Toll Assistance			
Services Calls			
- business	\$0.35		
- residential	\$0.35		

(B) Person-to-Person Per Call Charge

Rate Per Call \$4.00

(C) Station-to-Station Per Call Charges - Calling Card Option

Call Type	Rate Per Call			
Calling Card				
LEC Card				
Fully Automated	\$0.60			
Operator Assisted	\$1.60			
Operator Dialed	\$2.60			
Calling Card - Option 1				
Fully Automated	\$1.25			
Operator Assisted	\$1.95			
Operator Dialed	\$2.95			
Calling Card - Option 2				
Fully Automated	\$1.25			
Operator Assisted	\$1.95			
Operator Dialed	\$2.95			

Call Type	<b>Rate Per Call</b>		
Calling Card			
Calling Card - Option 3			
Operator Assisted	\$1.00		
Operator Dialed	\$2.00		
Calling Card - Option 4, Value Card Plus			
Operator Assisted	\$1.00		
Operator Dialed	\$1.50		

Issued: February 1, 2010

Donna Daniele, Area Manager 5130 hacienda Drive, Dublin California 94568

SBC Long Distance, LLC
d/b/a AT&T Long Distance
NY PSC Tariff No. 1 – Telephone

## PRICE LIST

- 1.1 Operator Toll Assistance Services (continued)
  - 1.1.2 Access Method Toll Free Access Number (continued)
    - (C) Station-to-Station Per Call Charges Calling Card Option (continued)

Call Type	Rate Per Call
Calling Card - Option 2 Categories	
Category 11	
Fully Automated	\$1.25
Operator Assisted	\$1.25
Operator Dialed	\$2.25
Category 12	
Operator Assisted	\$1.00
Operator Dialed	\$2.00

Call Type	<b>Rate Per Call</b>
Calling Card - Option 3 Categories	
Category 21	
Fully Automated	\$0.50
Operator Assisted	\$1.00
Operator Dialed	\$2.00

(D) Station-to-Station Per Call Charges

Collect, Third Number or Sent Paid

Call Type	Rate Per Call			
Collect				
Fully Automated	\$1.80			
Operator Assisted	\$1.80			
Operator Dialed	\$2.60			
Third Party				
Fully Automated	\$1.85			
Operator Assisted	\$1.85			
Operator Dialed	\$2.65			
Sent Paid				
Operator Assisted	\$1.85			
Operator Dialed	\$2.65			

#### 1.1.3 MRC

(A) Calling Card - Option 4, Value Card Plus

The MRC is \$1.95.

- 1.1.4 Busy Line Verify/Interrupt
  - (A) Verify charge per occurrence \$6.50
  - (B) Interrupt charge per occurrence \$6.50

Issued: March 25, 2009

SBC Long Distance, LLC
d/b/a AT&T Long Distance
NY PSC Tariff No. 1 – Telephone

## PRICE LIST

- 1.2 Directory Assistance Services
  - 1.2.1 Reserved for future use
  - 1.2.2 Call Completed Via All Other Access Methods

The rate for Directory Assistance Service is \$1.25 per call from a non-payphone and from a payphone. If the call originates from a payphone, the payphone originate charge described in Section 4.10.2 of the Tariff and Section 1.8.4 of the Price List may apply.

1.2.3 Directory Assistance Call Completion

The rate for Directory Assistance Call Completion is \$0.50 per call.

1.3 Reserved for future use.

# SBC Long Distance, LLC d/b/a AT&T Long Distance NY PSC Tariff No. 1 – Telephone

Attachment I 2<sup>nd</sup> Revised Page 6 Superseding 1<sup>st</sup> Revised Page 6

# PRICE LIST

# 1.4 Outbound Services-Switched Access

#### 1.4.1 MTS

	Peak		Off-Peak	
	Initial	Add'l	Initial	Add'l
	Period	Period	Period	Period
Business MTS	\$0.67	\$0.67	\$0.57	\$0.57
Rate Change Effective April 12, 2010 (T)	\$0.70(I)	\$0.70(I)	\$0.67(I)	\$0.67(I)
Residential MTS	\$0.33	\$0.33	\$0.33	\$0.33

1.4.2 Reserved for future use

Issued: February 1, 2010

Donna Daniele, Area Manager 208 South Akard Street, Dallas, Texas 75202

Т

SBC Long Distance, LLC
d/b/a AT&T Long Distance
NY PSC Tariff No. 1 – Telephone

Attachment I 1<sup>st</sup> Revised Page 7 Superseding Original Page 7

Effective: February 23, 2010

#### PRICE LIST

- 1.4 Outbound Services-Switched Access (continued)
  - **Consumer Outbound Services** 1.4.3
    - Long Distance II<sup>1</sup> (A)

The usage rate is \$0.10 per minute. For interstate MRC, see Section 4.4.3 (F) of the Company's interstate Voice Product Reference and Pricing Guide which may be found at www.att.com1.4 Outbound Services-Switched Access (continued)

60 Block of Time II<sup>3</sup> **(B)** 

> The MRC is \$5.00 per BTN until March 12, 2010 at which time the MRC will increase to \$6.00. The rate is \$0.07 per minute for all 1+ Direct-Dialed outbound T/I intrastate calls completed after the 60 minute block of time has been used.

500 Block of Time II<sup>3</sup> (C)

> The MRC is \$22.95 per BTN. The rate is \$0.07 per minute for all 1+ Direct-Dialed outbound intrastate calls completed after the 500 minute block of time has been used.

AT&T Worldwide & US Calling<sup>SM</sup> formerly known as JustCall<sup>SM</sup> Global<sup>4</sup> (D)

Customers selecting the AT&T Worldwide & US Calling<sup>SM</sup> plan, as set forth in Section 8.4.2 (AC) of Company's Voice Reference and Product Pricing Guidebook, will receive a rate of \$0.20 per minute for outbound 1+ Direct-Dialed (not including Calling Card calls) Intrastate long distance.

Simply Talk<sup>SM</sup> 5 Cents<sup>2</sup> (F)

> The usage rate is \$0.05 per minute. For Customers subscribing to this Service for the provision of interstate and intrastate calling, the interstate/intrastate MRC is \$5.00. For Customers subscribing to this Service for the provision of intrastate calling only, the MRC is \$5.00.

Simply Talk<sup>SM 2</sup> (H)

The MRC is \$1.00 and usage rate is \$0.25 per minute.

<sup>1</sup>This Service is no longer available to new Customers effective April 12, 2004.

<sup>2</sup>This Service is no longer available to new Customers or existing Customers at new locations effective July 1, 2005.

<sup>3</sup>This Service is no longer available to new Customers or existing Customers at new locations effective Februar 15, 2007.

<sup>4</sup>This Service is no longer available to new Customers or existing Customers at new locations effective October 22, 2008.

Issued: January 15, 2010

SBC Long Distance, LLC
d/b/a AT&T Long Distance
NY PSC Tariff No. 1 – Telephone

## PRICE LIST

- 1.4 Outbound Services-Switched Access (continued)
  - 1.4.4 Business Outbound Services
    - (A) Business Default Plan for Hierarchical Billing Switched

Switched	Peak		Off-Peak	
Switched	<b>Initial Period</b>	Initial Period Add'l Period		Add'l Period
InterLATA	\$0.0750	\$0.0150	\$0.0750	\$0.0150
IntraLATA	\$0.0750	\$0.0150	\$0.0750	\$0.0150

- 1.5 Outbound Services-Dedicated Access
  - 1.5.1 Business Default Plan for Hierarchical Billing Dedicated

Dedicated	<b>Initial Period</b>	Add'l Period
InterLATA	\$0.0650	\$0.0130
IntraLATA	\$0.0650	\$0.0130

Effective: March 12, 2010

SBC Long Distance, LLC	Attachment I
d/b/a AT&T Long Distance	2 <sup>nd</sup> Revised Page 9
NY PSC Tariff No. 1 – Telephone	Superseding 1 <sup>st</sup> Revised Page 9

#### PRICE LIST

#### 1.6 AT&T Toll Free Services

- 1.6.1 AT&T Toll Free Services-Switched (continued)
  - (A) Reserved for future use
  - (B) Business Toll Free Services
    - .1 Reserved for future use
    - .2 Reserved for future use
    - .3 Toll Free Business Default

	Peak		Off Peak	
	Initial	Addl	Initial	Addl
	Period	Period	Period	Period
	\$0.67	\$0.67	\$0.57	\$0.57
Rate Change Effective April 12, 2010 (T)	\$0.70(I)	\$0.70(I)	\$0.67(I)	\$0.67(I)

- 1.6.2 Reserved for future use.
- 1.6.3 Optional Feature Charges

The description and rates for available optional features may be found in the Company's interstate Voice Product Reference and Pricing Guide which may be found at www.att.com.guidebook

1.6.4 MRC

The MRC for a TFN terminating over a Switched Access arrangement may be found in the Company's Voice Produce Reference and Pricing Guide which may be found at www.att.com/guidebook.

1.6.5 Reserved for future use.

<sup>1</sup> Effective November 12, 2007, the dedicated service offering associated with Toll Free Service will no longer be available to new Customers. Existing term plan Customers may add, move, remove or change lines and/or locations for the duration of their current term plan agreement. Adds, moves and changes for this Service are no longer available to existing Customers upon expiration of the Customer's Term Plan Agreement or Customers currently on a month to month basis.

Issued: February 1, 2010

Donna Daniele, Area Manager 208 South Akard Street, Dallas, Texas 75202

SBC Long Distance, LLC	Attachment I
d/b/a AT&T Long Distance	Original Page 10
NY PSC Tariff No. 1 – Telephone	

- 1.6 AT&T Toll Free Services<sup>2</sup> (continued)
  - 1.6.6 AT&T Enhanced Toll Free Services<sup>2</sup>

For all billing options available to Customers subscribing to Enhanced Toll Free Service, the charges associated with High Volume Calling Business Optional Calling Plan selected by the Customer are specified in Section 1.7 of this Price List and are in addition to the feature charges described in Section 1.6.6 of this Price List. The interstate MRCs and one-time charges associated with the Toll Free Number may be found in the Company's interstate Voice Product Reference and Pricing Guide which may be found at www.att.com/guidebook.

(A) Combined Transport and Usage Billing<sup>1</sup>

<sup>1</sup> This billing option is no longer available to new Customers effective June 22, 2005.

The per minute feature charge is specified in the Section 1.7 of this Price List for the High Volume Calling Business Optional Calling Plan selected by the Customer in the section entitled "With CMR" or "With CTUB."

# 1.6 AT&T Toll Free Services<sup>2</sup> (continued)

- 1.6.6 AT&T Enhanced Toll Free Services<sup>2</sup> (continued)
  - (B) Per Minute Feature Billing

Feature	Rate Per Minute
CTS (call transfer, call transfer consult,	
call transfer conference, menu again unattended,	\$0.000
and menu again attended)	
Call Routing (all features)*	\$0.016
Busy/Ring No Answer Overflow	\$0.000
Origin Dependent Routing	\$0.000
Authorization Codes	\$0.000
Extension Routing	\$0.000
Play Announcement	\$0.000
Continuation of Business Announcements	\$0.000
Text-to-Speech	\$0.000
Standard Reports	\$0.000
Web Tool Access	\$0.000
Locator Services	\$0.000
Network Call Center Availability Routing	\$0.000
Network Queuing	\$0.000
Alternate Routing	\$0.000
DTMF Cut-Through Toggle	\$0.000
Menu Routing up to 2 Tier	\$0.000
Menu Routing n-Tier	\$0.000
Unlimited Storage Blocks	\$0.000
Speech Recognition	\$0.0041

\*All features listed with the exception of Speech Recognition are included with the usage rate of \$0.016 per minute. CTS< Locator Service, Network Call Center Availability Routing, Network Queuing and n-tier Menu Routing have NRC's and MRC's associated with them. See Section 1.6.6(A) of this Price List.

<sup>2</sup>This Service is no longer available for new Customer term plan agreements effective July 9, 2007. Existing Customers may add, move, remove, or change lines and/or locations for the duration of their current term plan agreement. This Service is no longer available to existing Customers upon expiration of the Customer's Term Plan.

<sup>1</sup>Effective November 12, 2007, the dedicated service offering associated with Toll Free Service will no longer be available to new Customers. Existing term plan Customers may add, move, remove or change lines and/or locations for the duration of their current term plan agreement. Adds, moves and changes for this Service are no longer available to existing Customers upon expiration of the Customer's Term Plan Agreement or Customers currently on a month to month basis.

### 1.6 AT&T Toll Free Services<sup>3</sup> (continued)

- 1.6.6 AT&T Enhanced Toll Free Services<sup>2</sup> (continued)
  - (C) Per Feature Billing

Feature	Per Feature Per Call
CTS (call transfer, call transfer consult, call transfer conference, menu again unattended, and menu again attended)	\$0.250
Call Routing	\$0.070
Busy/Ring No Answer Overflow	\$0.030
Origin Dependent Routing	\$0.070
Authorization Codes	\$0.020
Extension Routing	\$0.020
Play Announcement	\$0.070
Continuation of Business Announcements	\$0.070
Locator Services	\$0.050
Network Call Center Availability Routing	\$0.030
Network Queuing	\$0.300
Alternate Routing	
DTMF Cut-Through Toggle	
Menu Routing up to 2 Tier	
Menu Routing n-Tier	No per call
Unlimited Storage Blocks	feature charge
Text-to-Speech	
Standard Reports	
Web Tool Access	

Minimum/Maximum Charge	Per Call
Minimum	\$0.015
Maximum	\$0.500

<sup>2</sup>This Service is no longer available for new Customer term plan agreements effective July 9, 2007. Existing Customers may add, move, remove, or change lines and/or locations for the duration of their current term plan agreement. This Service is no longer available to existing Customers upon expiration of the Customer's Term Plan.

<sup>3</sup>Effective November 12, 2007, the dedicated service offering associated with Toll Free Service will no longer be available to new Customers. Existing term plan Customers may add, move, remove or change lines and/or locations for the duration of their current term plan agreement. Adds, moves and changes for this Service are no longer available to existing Customers upon expiration of the Customer's Term Plan Agreement or Customers currently on a month to month basis.

Effective: April 27, 2009

## PRICE LIST

# 1.6 AT&T Toll Free Services<sup>3</sup> (continued)

- 1.6.6 AT&T Enhanced Toll Free Services<sup>2</sup> (continued)
  - (D) Enhanced Per Feature Billing

Feature	Per Feature
	Per Call
Call Routing	\$0.070
Busy/Ring No Answer Overflow	\$0.030
Origin Dependent Routing	\$0.070
Authorization Codes	\$0.020
Extension Routing	\$0.020
Play Announcement	\$0.070
Continuation of Business Announcements	\$0.070
Locator Services	\$0.050
Call Transfer - Redirection Attempt	\$0.400
Call Transfer - Completed Call	\$0.070
Network Call Center Availability Routing	\$0.030
Network Queuing	\$0.300
Alternate Routing	
DTMF Cut-Through Toggle	No per call
Menu Routing up to 2 Tier	feature charge
Menu Routing n-Tier	reature entarge

Minimum/Maximum Charge	Per Call
Minimum	\$0.015
Maximum	\$0.500

<sup>2</sup>This Service is no longer available for new Customer term plan agreements effective July 9, 2007. Existing Customers may add, move, remove, or change lines and/or locations for the duration of their current term plan agreement. This Service is no longer available to existing Customers upon expiration of the Customer's Term Plan.

<sup>3</sup>Effective November 12, 2007, the dedicated service offering associated with Toll Free Service will no longer be available to new Customers. Existing term plan Customers may add, move, remove or change lines and/or locations for the duration of their current term plan agreement. Adds, moves and changes for this Service are no longer available to existing Customers upon expiration of the Customer's Term Plan Agreement or Customers currently on a month to month basis.

SBC Long Distance, LLC
d/b/a AT&T Long Distance
NY PSC Tariff No. 1 – Telephone

#### 1.7 Custom Business Services

1.7.1 High Volume Calling<sup>1</sup>

<sup>1</sup> This Service is no longer available to new Customers or existing Customers at new locations effective August 1, 2003.

- (A) Outbound Calls
  - .1 The per minute usage rate for InterLATA calls is as follows:

		Per Minute Rate				
MMC	MAC	МТМ	1-Year	2-Year	3-Year	
			Term Plan	Term Plan	<b>Term Plan</b>	
\$50	\$600	\$0.0890	\$0.0820	\$0.0780	\$0.0740	
\$200	\$2,400	\$0.0890	\$0.0800	\$0.0760	\$0.0720	
\$500	\$6,000	\$0.0880	\$0.0780	\$0.0740	\$0.0700	
\$1,000	\$12,000	\$0.0860	\$0.0760	\$0.0720	\$0.0680	
\$2,500	\$30,000	\$0.0840	\$0.0740	\$0.0700	\$0.0660	
\$5,000	\$60,000	\$0.0820	\$0.0720	\$0.0680	\$0.0640	
\$10,000	\$120,000	\$0.0800	\$0.0700	\$0.0660	\$0.0620	
\$15,000	\$180,000	\$0.0780	\$0.0680	\$0.0650	\$0.0610	
\$20,000	\$240,000	\$0.0760	\$0.0660	\$0.0640	\$0.0600	

The per minute usage rate for IntraLATA calls is as follows:

		Per Minute Rate				
MMC	MAC	MTM	1-Year	2-Year	3-Year	
			<b>Term Plan</b>	Term Plan	Term Plan	
\$50	\$600	\$0.0890	\$0.0820	\$0.0780	\$0.0740	
\$200	\$2,400	\$0.0890	\$0.0800	\$0.0760	\$0.0720	
\$500	\$6,000	\$0.0880	\$0.0780	\$0.0740	\$0.0700	
\$1,000	\$12,000	\$0.0860	\$0.0760	\$0.0720	\$0.0680	
\$2,500	\$30,000	\$0.0840	\$0.0740	\$0.0700	\$0.0660	
\$5,000	\$60,000	\$0.0820	\$0.0720	\$0.0680	\$0.0640	
\$10,000	\$120,000	\$0.0800	\$0.0700	\$0.0660	\$0.0620	
\$15,000	\$180,000	\$0.0780	\$0.0680	\$0.0650	\$0.0610	
\$20,000	\$240,000	\$0.0760	\$0.0660	\$0.0640	\$0.0600	

SBC Long Distance, LLC
d/b/a AT&T Long Distance
NY PSC Tariff No. 1 – Telephone

- 1.7 Custom Business Services (continued)
  - 1.7.1 High Volume Calling<sup>1</sup> (continued)

<sup>1</sup> This Service is no longer available to new Customers or existing Customers at new locations effective August 1, 2003.

- (A) Outbound Calls (continued)
  - .2 High Volume Dedicated Outbound Calling

The per minute usage rates for InterLATA calls are as follows.

		Per Minute Rate			
MMC	MAC	МТМ	1-Year	2-Year	3-Year
			Term Plan	Term Plan	<b>Term Plan</b>
\$50	\$600	\$0.0710	\$0.0640	\$0.0600	\$0.0560
\$200	\$2,400	\$0.0710	\$0.0620	\$0.0580	\$0.0540
\$500	\$6,000	\$0.0700	\$0.0600	\$0.0560	\$0.0520
\$1,000	\$12,000	\$0.0680	\$0.0580	\$0.0540	\$0.0500
\$2,500	\$30,000	\$0.0660	\$0.0560	\$0.0520	\$0.0480
\$5,000	\$60,000	\$0.0640	\$0.0540	\$0.0500	\$0.0460
\$10,000	\$120,000	\$0.0620	\$0.0520	\$0.0480	\$0.0440
\$15,000	\$180,000	\$0.0600	\$0.0500	\$0.0470	\$0.0430
\$20,000	\$240,000	\$0.0580	\$0.0480	\$0.0460	\$0.0420

The per minute usage rates for IntraLATA calls are as follows.

		Per Minute Rate				
MMC	MAC	MTM	1-Year Term Plan	2-Year Term Plan	3-Year Term Plan	
\$50	\$600	\$0.0710	\$0.0640	\$0.0600	\$0.0560	
\$200	\$2,400	\$0.0710	\$0.0620	\$0.0580	\$0.0540	
\$500	\$6,000	\$0.0700	\$0.0600	\$0.0560	\$0.0520	
\$1,000	\$12,000	\$0.0680	\$0.0580	\$0.0540	\$0.0500	
\$2,500	\$30,000	\$0.0660	\$0.0560	\$0.0520	\$0.0480	
\$5,000	\$60,000	\$0.0640	\$0.0540	\$0.0500	\$0.0460	
\$10,000	\$120,000	\$0.0620	\$0.0520	\$0.0480	\$0.0440	
\$15,000	\$180,000	\$0.0600	\$0.0500	\$0.0470	\$0.0430	
\$20,000	\$240,000	\$0.0580	\$0.0480	\$0.0460	\$0.0420	

SBC Long Distance, LLC
d/b/a AT&T Long Distance
NY PSC Tariff No. 1 – Telephone

#### 1.7 Custom Business Services (continued)

1.7.1 High Volume Calling<sup>1</sup> (continued)

<sup>1</sup> This Service is no longer available to new Customers or existing Customers at new locations effective August 1, 2003.

- (B) Inbound Toll Free Calls
  - .1 High Volume Toll Free Calling Usage Rates
    - .a Without CMR

The per minute usage rates are the same as Section 1.7.1 (A).1 of the Price List.

.b With CMR

The rate is \$.03 per minute which applies in addition to the per minute usage rates in Section 1.7.1 (A).1 of the Price List.

- .2 High Volume Dedicated Toll Free Calling Usage Rates
  - .a Without CMR

The per minute usage rates are the same as Section 1.7.1 (A).2 of the Price List.

.b With CMR

The rate is \$.03 per minute which applies in addition to the per minute usage rates in Section 1.7.1 (A).2 of the Price List.

.3 Optional Feature Charges

The description and rates for available optional features may be found in the Company's interstate Voice Product Reference and Pricing Guide which may be found at <u>www.att.com/guidebook</u>.

SBC Long Distance, LLC d/b/a AT&T Long Distance <u>NY PSC Tariff No. 1 – Telephone</u> Attachment I 1<sup>st</sup> Revised Page 17 Superseding Original Page 17

#### PRICE LIST

- 1.7 Custom Business Services (continued)
  - 1.7.2 AT&T High Volume Calling  $II^2$ 
    - (A) Outbound Calls
      - .1 AT&T High Volume Outbound Calling II
        - .a The per minute usage rates for intrastate InterLATA calls are as follows:

			Per Minute Rate			
MMC	MAC	МТМ	1-Year	2-Year	3-Year	
			<b>Term Plan</b>	Term Plan	Term Plan	
\$50	\$600	\$0.1068(I)	\$0.0820	\$0.0780	\$0.0740	
\$200	\$2,400	\$0.1068(I)	\$0.0800	\$0.0760	\$0.0720	
\$500	\$6,000	\$0.1056(I)	\$0.0780	\$0.0740	\$0.0700	
\$1,000	\$12,000	\$0.1032(I)	\$0.0760	\$0.0720	\$0.0680	
\$2,500	\$30,000 <sup>1</sup>	\$0.1008(I)	\$0.0740	\$0.0700	\$0.0660	
\$5,000	$$60,000^{1}$	\$0.0984(I)	\$0.0720	\$0.0680	\$0.0640	
\$10,000	\$120,000 <sup>1</sup>	\$0.0960(I)	\$0.0700	\$0.0660	\$0.0620	
\$15,000	\$180,000 <sup>1</sup>	\$0.0936(I)	\$0.0680	\$0.0650	\$0.0610	
\$20,000	\$240,000 <sup>1</sup>	\$0.0912(I)	\$0.0660	\$0.0640	\$0.0600	

<sup>1</sup>This Service is no longer available to for new Customer term agreements effective March 1, 2007. Existing Customers may add, move, remove, or change lines and/or locations for the duration of their current term agreement.

<sup>2</sup> Effective November 12, 2007, the Dedicated Service Offering associated with High Volume Calling will no longer be available to new Customers.

Issued: May 27, 2009

SBC Long Distance, LLC
d/b/a AT&T Long Distance
NY PSC Tariff No. 1 – Telephone

Attachment I 2<sup>nd</sup> Revised Page 18 Superseding 1<sup>st</sup> Revised Page 18

Effective: February 12, 2010

### PRICE LIST

- 1.7 Custom Business Services (continued)
  - 1.7.2 AT&T High Volume Calling II<sup>2</sup>(continued)
    - (A) Outbound Calls (continued)
      - .1 AT&T High Volume Outbound Calling II (continued)
        - .b The per minute usage rates for intrastate IntraLATA calls are as follows:

		Per Minute Rate				
MMC	MAC	мтм	1-Year	2-Year	3-Year	
		TALITAL	Term Plan	Term Plan	Term Plan	
\$50	\$600	\$0.1068	\$0.0820	\$0.0780	\$0.0740	
\$200	\$2,400	\$0.1068	\$0.0800	\$0.0760	\$0.0720	
\$500	\$6,000	\$0.1056	\$0.0780	\$0.0740	\$0.0700	
\$1,000	\$12,000	\$0.1032	\$0.0760	\$0.0720	\$0.0680	
\$2,500	\$30,000 <sup>1</sup>	\$0.1008	\$0.0740	\$0.0700	\$0.0660	
\$5,000	$$60,000^{1}$	\$0.0984	\$0.0720	\$0.0680	\$0.0640	
\$10,000	\$120,000 <sup>1</sup>	\$0.0960	\$0.0700	\$0.0660	\$0.0620	
\$15,000	\$180,000 <sup>1</sup>	\$0.0936	\$0.0680	\$0.0650	\$0.0610	
\$20,000	\$240,000 <sup>1</sup>	\$0.0912	\$0.0660	\$0.0640	\$0.0600	

.c Out Of Term

Effective August 18, 2008 the Out Of Term per minute usage rates for intrastate InterLATA and intrastate IntraLATA calls associated with 1-Year, 2-Year and 3-Year Term Plans are as follows:

OUT OF TERM					
MAC	Per Minute Rate				
\$600	\$0.1100(I)				
\$2,400	\$0.1080(I)				
\$6,000	\$0.1040(I)				
\$12,000	\$0.1010(I)				
\$30,000	\$0.0980(I)				
\$60,000	\$0.0960(I)				
\$120,000	\$0.0920(I)				
\$180,000	\$0.0910(I)				
\$240,000	\$0.0900(I)				

<sup>1</sup>This Service is no longer available to for new Customer term agreements effective March 1, 2007. Existing Customers may add, move, remove, or change lines and/or locations for the duration of their current term agreement.

<sup>2</sup> Effective November 12, 2007, the Dedicated Service Offering associated with High Volume Calling will no longer be available to new Customers.

Issued: January 4, 2010

Donna Daniele, Area Manager 208 South Akard Street, Dallas, Texas 75202

SBC Long Distance, LLC	
d/b/a AT&T Long Distance	
NY PSC Tariff No. 1 – Telephone	

Attachment I 1<sup>st</sup> Revised Page 19 Superseding Original Page 19

## PRICE LIST

- 1.7 Custom Business Services (continued)
  - 1.7.2 AT&T High Volume Calling II<sup>2</sup>(continued)
    - (A) Outbound Calls (continued)
      - .2 AT&T High Volume Dedicated Outbound Calling II
        - .a The per minute usage rates for InterLATA calls are as follows.

		Per Minute Rate				
MMC	MAC	МТМ	1-Year	2-Year	3-Year	
			Term Plan	Term Plan	<b>Term Plan</b>	
\$50	\$600	\$0.0852(I)	\$0.0640	\$0.0600	\$0.0560	
\$200	\$2,400	\$0.0852(I)	\$0.0620	\$0.0580	\$0.0540	
\$500	\$6,000	\$0.0840(I)	\$0.0600	\$0.0560	\$0.0520	
\$1,000	\$12,000	\$0.0816(I)	\$0.0580	\$0.0540	\$0.0500	
\$2,500	\$30,000 <sup>1</sup>	\$0.0792(I)	\$0.0560	\$0.0520	\$0.0480	
\$5,000	$60,000^{1}$	\$0.0768(I)	\$0.0540	\$0.0500	\$0.0460	
\$10,000	\$120,000 <sup>1</sup>	\$0.0744(I)	\$0.0520	\$0.0480	\$0.0440	
\$15,000	\$180,000 <sup>1</sup>	\$0.0720(I)	\$0.0500	\$0.0470	\$0.0430	
\$20,000	\$240,000 <sup>1</sup>	\$0.0696(I)	\$0.0480	\$0.0460	\$0.0420	

<sup>1</sup>This Service is no longer available to for new Customer term agreements effective March 1, 2007. Existing Customers may add, move, remove, or change lines and/or locations for the duration of their current term agreement.

<sup>2</sup> Effective November 12, 2007, the Dedicated Service Offering associated with High Volume Calling will no longer be available to new Customers.

SBC Long Distance, LLC
d/b/a AT&T Long Distance
NY PSC Tariff No. 1 – Telephone

Attachment I 2<sup>nd</sup> Revised Page 20 Superseding 1<sup>st</sup> Revised Page 20

Effective: February 12, 2010

### PRICE LIST

- 1.7 Custom Business Services (continued)
  - 1.7.2 AT&T High Volume Calling II<sup>2</sup> (continued)
    - (A) Outbound Calls (continued)
      - .2 AT&T High Volume Dedicated Outbound Calling II (continued)
        - .b The per minute usage rates for IntraLATA calls are as follows.

		Per Minute Rate			
MMC	MAC	МТМ	1-Year	2-Year	3-Year
			Term Plan	Term Plan	Term Plan
\$50	\$600	\$0.0852	\$0.0640	\$0.0600	\$0.0560
\$200	\$2,400	\$0.0852	\$0.0620	\$0.0580	\$0.0540
\$500	\$6,000	\$0.0840	\$0.0600	\$0.0560	\$0.0520
\$1,000	\$12,000	\$0.0816	\$0.0580	\$0.0540	\$0.0500
\$2,500	\$30,000 <sup>1</sup>	\$0.0792	\$0.0560	\$0.0520	\$0.0480
\$5,000	$$60,000^{1}$	\$0.0768	\$0.0540	\$0.0500	\$0.0460
\$10,000	\$120,000 <sup>1</sup>	\$0.0744	\$0.0520	\$0.0480	\$0.0440
\$15,000	\$180,000 <sup>1</sup>	\$0.0720	\$0.0500	\$0.0470	\$0.0430
\$20,000	\$240,000 <sup>1</sup>	\$0.0696	\$0.0480	\$0.0460	\$0.0420

### .c Out Of Term

Effective August 18, 2008 the Out Of Term per minute usage rates for intrastate InterLATA and intrastate IntraLATA calls associated with 1-Year, 2-Year and 3-Year Term Plans are as follows:

OUT OF TERM			
MAC Per Minute Rate			
\$600	\$ 0.0830(I)		
\$2,400	\$ 0.0800(I)		
\$6,000	\$ 0.0770(I)		
\$12,000	\$ 0.0740(I)		
\$30,000	\$ 0.0710(I)		
\$60,000	\$ 0.0680(I)		
\$120,000	\$ 0.0660(I)		
\$180,000	\$ 0.0640(I)		
\$240,000	\$ 0.0620(I)		

<sup>1</sup>This Service is no longer available to for new Customer term agreements effective March 1, 2007. Existing Customers may add, move, remove, or change lines and/or locations for the duration of their current term agreement.

<sup>2</sup> Effective November 12, 2007, the Dedicated Service Offering associated with High Volume Calling will no longer be available to new Customers.

Issued: January 4, 2010

Donna Daniele, Area Manager 208 South Akard Street, Dallas, Texas 75202

SBC Long Distance, LLC
d/b/a AT&T Long Distance
NY PSC Tariff No. 1 – Telephone

Effective: April 27, 2009

### PRICE LIST

#### 1.7 Custom Business Services (continued)

- 1.7.2 AT&T High Volume Calling II <sup>2</sup>(continued)
  - (B) Inbound Toll Free Calls
    - .1 AT&T High Volume Toll Free Calling II Usage Rates
      - .a Without CMR

The per minute usage rates are the same as Section 1.7.2 (A).1 of this Price List.

.b With CMR

The rate is \$.03 per minute which applies in addition to the per minute usage rates in Section 1.7.2 (A).1 of this Price List.

- .2 AT&T High Volume Dedicated Toll Free Calling II Usage Rates
  - .a Without CMR

The per minute usage rates are the same as Section 1.7.2 (A).2 of this Price List.

.b With CMR

The rate is \$.03 per minute which applies in addition to the per minute usage rates in Section 1.7.2 (A).2 of this Price List.

.3 Optional Feature Charges

The description and rates for available optional features may be found in the Company's interstate Voice Product Reference and Pricing Guide which may be found at www.sbc.com.

- 1.7.3 Reserved for future use
- 1.7.4 Reserved for future use

<sup>2</sup> Effective November 12, 2007, the Dedicated Service Offering associated with High Volume Calling will no longer be available to new Customers.

SBC Long Distance, LLC	Attachment I
d/b/a AT&T Long Distance	2 <sup>nd</sup> Revised Page 22
NY PSC Tariff No. 1 – Telephone	Supeprseding 1 <sup>st</sup> Revised Page 22

#### 1.7 Custom Business Services (continued)

1.7.5 Business Long Distance 75<sup>1</sup>

<sup>1</sup>This Service is no longer available to new Customers or to existing Customers at new locations effective February 1, 2001.

<b>Customer Commitment</b>	<b>Rate Per Minute</b>
Month-to-Month	\$0.0890
1 Year Term Plan	\$0.0820

1.7.6 Business Long Distance  $200^2$ 

<sup>2</sup>This Service is no longer available to new Customers effective June 3, 2002.

The per minute usage rates for outbound and TFS calls are as follows:

<b>Customer Commitment</b>	<b>Rate Per Minute</b>
Month-to-Month	\$0.0890
1 Year Term Plan	\$0.0800

For fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, the usage rate is \$0.14 per minute. The per call charge may be found in Section 1.1.1 (B).2.a, Section 1.1.2 (B), and Section 1.1.2 (C) of this Price List.

1.7.7 AT&T Business Calling formerly Long Distance for Business

The MRC is \$1.25 and the usage rate is \$0.53 per minute for outbound and TFS calls T until April 12, 2010 at which time the usage rate will increase to \$0.555. For fully T/I automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, the usage rate is \$0.53 per minute until April 12, 2010 at which time the usage rate will T increase to \$0.555. The per call charge may be found in Section 1.1.1 (B).2.a, Section T/I 1.1.2 (B), and Section 1.1.2 (C) of this Price List.

1.7.8 Total Solutions Plus<sup>1</sup>

<sup>1</sup>This service is no longer available to new Customers or existing Customers at new locations effective April 1, 2004.

The usage rate is \$0.10 per minute for outbound and TFS calls. For fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, the usage rate is \$0.15 per minute. The per call charge may be found in Section 1.1.1 (B).2.a, Section 1.1.2 (B), and Section 1.1.2 (C) of this Price List.

Effective: March 12, 2010

Issued: February 1, 2010

Attachment I
2 <sup>nd</sup> Revised Page 23
Superseding 1 <sup>st</sup> Revised Page 23

- 1.7 Custom Business Services (continued)
  - 1.7.9 Business Long Distance  $50^2$

<sup>2</sup>This Service is no longer available to new Customers or existing Customers at new locations effective February 12, 2007

The per minute usage rates for outbound and TFS calls are as follows:

<b>Customer Commitment</b>	<b>Rate Per Minute</b>
Month-to-Month	\$0.0800
1 Year Term Plan	\$0.0800
2 Year Term Plan	\$0.0800

For fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, category 11, the usage rate is \$0.15 per minute. The per call charge may be found in Section 1.1.1 (B).2.a, Section 1.1.2 (B), and Section 1.1.2 (C) of this Price List.

1.7.10 AT&T Business Calling \$5.95<sup>1</sup> formerly Business Domestic Saver

The usage rate is \$0.09 per minute until January 12, 2010 at which time the per minuteTusage rate will increase to \$0.12 for outbound and TFS calls. For fully automated,T/Ioperator assisted, and operator dialed calls billed to the Calling Card - Option 2, categoryT/I11, the usage rate is \$0.16 per minute until January 12, 2010 at which time the per minuteTusage rate will increase to \$0.20. The per call charge may be found in Section 1.1.1T/I(B).2.a, Section 1.1.2 (B), and Section 1.1.2 (C) of this Price List.T

1.7.11 AT&T Business Calling \$15 formerly Business Domestic Saver 15

The per minute usage rates for outbound and TFS calls are as follows:

Customer Commitment	<b>Rate Per Minute</b>
Month-to-Month	\$0.0800
Rate Change Effective January 12, 2010 (T)	\$0.1100(I)
1 Year Term Plan	\$0.0800
2 Year Term Plan	\$0.0800

For fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, category 11, the usage rate is \$0.15 per minute until January 12, 2010 at which time the per minute usage rate will increase to \$0.20. The per call charge may be found in Section 1.1.1 (B).2.a, Section 1.1.2 (B), and Section 1.1.2 (C) of this Price List.

<sup>1</sup>This service is no longer available to new Customers or existing Customers at new locations effective May 12, 2009.

Т

T/I

Issued: November 4, 2009

SBC Long Distance, LLC
d/b/a AT&T Long Distance
NY PSC Tariff No. 1 – Telephone

Effective: April 27, 2009

#### PRICE LIST

- 1.7 Custom Business Services (continued)
  - 1.7.12 SBC Long Distance Virtual Private Network (VPN)<sup>1</sup>

<sup>1</sup>This Service is no longer available for new Customer term plan agreements effective July 15, 2006. Existing Customers may add, move, remove or change lines and/or locations for the duration of their current term plan agreement. This Service is no longer available to existing Customers upon expiration of the Customer's term plan.

(A) Usage Rates

1-Year 2-Year 3-Year **Term Plan Term Plan Term Plan** Call Rate Type A InterLATA \$0.1100 \$0.1140 \$0.1120 IntraLATA \$0.0880 \$0.0860 \$0.0840 Call Rate Type B InterLATA \$0.0660 \$0.0640 \$0.0620 IntraLATA \$0.0660 \$0.0640 \$0.0620 Call Rate Type C InterLATA \$0.0560 \$0.0530 \$0.0500 IntraLATA \$0.0560 \$0.0530 \$0.0500

The per minute usage rates are as follows:

(B) Per Call Charges

For remote access calls, a per call charge of \$0.25 applies in addition to the usage charge shown in Section 1.7.12 (A) of this Price List.

Effective: April 27, 2009

### PRICE LIST

#### 1.7 Custom Business Services (continued)

1.7.12 SBC Long Distance Virtual Private Network (VPN)<sup>1</sup> (continued)

<sup>1</sup>This Service is no longer available for new Customer term plan agreements effective July 15, 2006. Existing Customers may add, move, remove or change lines and/or locations for the duration of their current term plan agreement. This Service is no longer available to existing Customers upon expiration of the Customer's term plan.

(C) Feature Charges

The OTCs and MRCs shown below are in addition to the usage charges set forth in Section 1.7.12 (A) of this Tariff and the per call charges set forth in Section 1.7.12 (B) of this Tariff.

	OTC	MRC
Network Overflow		
- Set up charge	\$50 per primary switch/trunk group	
- Change request charge	\$50 per switch/trunk group	
- Cancellation charges	\$50 per order	
VPN Authorization Code		\$30 per
		100 codes
- Set up charge	\$50 per 100 codes	
- Change request charge	\$50 per block up to 100 codes	
- Order cancellation charge	\$50 per occurrence	
- Feature cancellation charge	\$50 per occurrence	
Call Screen Routing		\$150 per VPN
- Initial set up & design	\$500 per VPN	
- Major change charge	\$500 per change order	
- Minor change charge	\$50 per change order	
- Cancellation charge	\$500 per VPN	

- 1.7 Custom Business Services (continued)
  - 1.7.13 Business Long Distance  $100^2$

The per minute usage rates for outbound and TFS calls are as follows:

<b>Customer Commitment</b>	<b>Rate Per Minute</b>
Month-to-Month	\$0.0800
1 Year Term Plan	\$0.0800
2 Year Term Plan	\$0.0800

For fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, category 11, the usage rate is \$0.14 per minute. The per call charge may be found in Section 1.1.1 (B).2.a, Section 1.1.2 (B), and Section 1.1.2 (C) of this Price List.

1.7.14 Business Block of Time 200<sup>1</sup>

<sup>1</sup>This service is no longer available to new Customers or existing Customers at new locations effective April 1, 2004.

The MRC is \$15.00 per BTN for a 200 minute block of time as described in Section 3.7.14 of this Tariff. For Direct-Dialed TFS and outbound calls, the rate is \$0.27 per minute for calls completed after the 200 minute block of time has been used. For fully automated, operator assisted, and operator dialed calling card calls billed to the Calling Card - Option 2, category 11, the rate is \$0.15 per minute. The per call charge may be found in Section 1.1.1 (B).2.a, Section 1.1.2 (B), and Section 1.1.2 (C) of this Price List.

1.7.15 Business Block of Time  $400^1$ 

<sup>1</sup>This service is no longer available to new Customers or existing Customers at new locations effective April 1, 2004.

The MRC is \$30.00 per BTN for a 400 minute block of time as described in Section 3.7.15 of this Tariff. For Direct-Dialed TFS and outbound calls, the rate is \$0.27 per minute for calls completed after the 400 minute block of time has been used. For fully automated, operator assisted, and operator dialed calling card calls billed to the Calling Card - Option 2, category 11, the rate is \$0.15 per minute. The per call charge may be found in Section 1.1.1 (B).2.a, Section 1.1.2 (B), and Section 1.1.2 (C) of this Price List.

<sup>2</sup>This Service is no longer available to new Customers or existing Customers at new locations effective February 12, 2007

SBC Long Distance, LLC
d/b/a AT&T Long Distance
NY PSC Tariff No. 1 – Telephone

Effective: April 27, 2009

#### PRICE LIST

- 1.7 Custom Business Services (continued)
  - 1.7.16 Business Domestic Saver Deluxe<sup>1</sup>

<sup>1</sup>This service is no longer available to new Customers or existing Customers at new locations effective April 1, 2004.

The usage rate is \$0.0800 per minute for outbound and TFS calls. For fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, category 11, the usage rate is \$0.16 per minute. The per call charge may be found in Section 1.1.1 (B).2.a, Section 1.1.2 (B), and Section 1.1.2 (C) of this Price List.

1.7.17 Business Domestic Saver 15 Deluxe<sup>1</sup>

<sup>1</sup>This service is no longer available to new Customers or existing Customers at new locations effective April 1, 2004.

The per minute usage rates for outbound and TFS calls are as follows:

<b>Customer Commitment</b>	<b>Rate Per Minute</b>
Month-to-Month	\$0.0900
1 Year Term Plan	\$0.0820
2 Year Term Plan	\$0.0820

For fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, category 11, the usage rate is \$0.15 per minute. The per call charge may be found in Section 1.1.1 (B).2.a, Section 1.1.2 (B), and Section 1.1.2 (C) of this Price List.

1.7.18 Business Domestic Saver 15 Connections 3 Service<sup>1</sup>

<sup>1</sup>This service is no longer available to new Customers or existing Customers at new locations effective April 1, 2004.

The per minute usage rates for outbound and TFS calls are as follows:

<b>Customer Commitment</b>	<b>Rate Per Minute</b>
1 Year Term Plan	\$0.0820
2 Year Term Plan	\$0.0820

For fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, the usage rate is \$0.15 per minute. The per call charge may be found in Section 1.1.1 (B).2.a, Section 1.1.2 (B), and Section 1.1.2 (C) of this Price List.

SBC Long Distance, LLC	
d/b/a AT&T Long Distance	
NY PSC Tariff No. 1 – Telephone	

Effective: April 27, 2009

### PRICE LIST

- 1.7 Custom Business Services (continued)
  - 1.7.19 Business Long Distance 50 Connections 3 Service<sup>1</sup>

<sup>1</sup>This service is no longer available to new Customers or existing Customers at new locations effective April 1, 2004.

The per minute usage rates for outbound and TFS calls are as follows:

<b>Customer Commitment</b>	<b>Rate Per Minute</b>
1 Year Term Plan	\$0.0820
2 Year Term Plan	\$0.0820

For fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, the usage rate is \$0.15 per minute. The per call charge may be found in Section 1.1.1 (B).2.a, Section 1.1.2 (B), and Section 1.1.2 (C) of this Price List.

1.7.20 Business Long Distance 100 Connections 3 Service<sup>1</sup>

<sup>1</sup>This service is no longer available to new Customers or existing Customers at new locations effective April 1, 2004.

The per minute usage rates for outbound and TFS calls are as follows:

<b>Customer Commitment</b>	<b>Rate Per Minute</b>
1 Year Term Plan	\$0.0800
2 Year Term Plan	\$0.0800

For fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, the usage rate is \$0.14 per minute. The per call charge may be found in Section 1.1.1 (B).2.a, Section 1.1.2 (B), and Section 1.1.2 (C) of this Price List.

1.7.21 Business Domestic Saver 15 Connections 2 Service<sup>2</sup>

<sup>2</sup>This Service is no longer available to new Customers or existing Customers at new locations effective February 12, 2007

The per minute usage rates for outbound and TFS calls are as follows:

<b>Customer Commitment</b>	<b>Rate Per Minute</b>
1 Year Term Plan	\$0.0680
2 Year Term Plan	\$0.0680

For fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, category 11, the usage rate is \$0.15 per minute. The per call charge may be found in Section 1.1.1 (B).2.a, Section 1.1.2 (B), and Section 1.1.2 (C) of this Price List.

```
Issued: March 25, 2009
```

SBC Long Distance, LLC d/b/a AT&T Long Distance <u>NY PSC Tariff No. 1 – Telephone</u> Attachment I Original Page 29

## PRICE LIST

- 1.7 Custom Business Services (continued)
  - 1.7.22 Business Long Distance 50 Connections 2 Service<sup>2</sup>

<sup>2</sup>This Service is no longer available to new Customers or existing Customers at new locations effective February 12, 2007

The per minute usage rates for outbound and TFS calls are as follows:

<b>Customer Commitment</b>	<b>Rate Per Minute</b>
1 Year Term Plan	\$0.0680
2 Year Term Plan	\$0.0680

For fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, category 11, the usage rate is \$0.15 per minute. The per call charge may be found in Section 1.1.1 (B).2.a, Section 1.1.2 (B), and Section 1.1.2 (C) of this Price List.

SBC Long Distance, LLC	
d/b/a AT&T Long Distance	
NY PSC Tariff No. 1 – Telephone	

Effective: April 27, 2009

### PRICE LIST

- 1.7 Custom Business Services (continued)
  - 1.7.23 Business Long Distance 100 Connections 2 Service<sup>1</sup>

The per minute usage rates for outbound and TFS calls are as follows:

<b>Customer Commitment</b>	<b>Rate Per Minute</b>
1 Year Term Plan	\$0.0680
2 Year Term Plan	\$0.0680

For fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, category 11, the usage rate is \$0.14 per minute. The per call charge may be found in Section 1.1.1 (B).2.a, Section 1.1.2 (B), and Section 1.1.2 (C) of this Price List.

1.7.24 Business Domestic Saver 15 Connections 1 Service<sup>1</sup>

The per minute usage rates for outbound and TFS calls are as follows:

<b>Customer Commitment</b>	<b>Rate Per Minute</b>
1 Year Term Plan	\$0.0680
2 Year Term Plan	\$0.0680

For fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, category 11, the usage rate is \$0.15 per minute. The per call charge may be found in Section 1.1.1 (B).2.a, Section 1.1.2 (B), and Section 1.1.2 (C) of this Price List.

1.7.25 Business Long Distance 50 Connections 1 Service<sup>1</sup>

The per minute usage rates for outbound and TFS calls are as follows:

<b>Customer Commitment</b>	<b>Rate Per Minute</b>
1 Year Term Plan	\$0.0680
2 Year Term Plan	\$0.0680

For fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, category 11, the usage rate is \$0.15 per minute. The per call charge may be found in Section 1.1.1 (B).2.a, Section 1.1.2 (B), and Section 1.1.2 (C) of this Price List.

1.7.26 Business Long Distance 100 Connections 1 Service<sup>1</sup>

The per minute usage rates for outbound and TFS calls are as follows:

<b>Customer Commitment</b>	<b>Rate Per Minute</b>
1 Year Term Plan	\$0.0680
2 Year Term Plan	\$0.0680

For fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, category 11, the usage rate is \$0.14 per minute. The per call charge may be found in Section 1.1.1 (B).2.a, Section 1.1.2 (B), and Section 1.1.2 (C) of this Price List.

<sup>1</sup>This Service is no longer available to new Customers or existing Customers at new locations effective February 12, 2007

SBC Long Distance, LLC
d/b/a AT&T Long Distance
NY PSC Tariff No. 1 – Telephone

Effective: April 27, 2009

## PRICE LIST

- 1.7 Custom Business Services (continued)
  - 1.7.27 AT&T High Volume Calling II Plus<sup>1</sup>

<sup>1</sup> This Service is no longer available for new Customer term plan agreements effective June 27, 2005. Existing Customers may add, move, remove or change lines and/or locations for the duration of their current term plan agreement. This Service is no longer available to existing Customers upon expiration of the Customer's Term Plan.

- (A) Outbound Calls
  - .1 H AT&T igh Volume Outbound Calling II Plus

The per minute usage rates are as follows:

MAC	1 Year	2 Year	3 Year
MAC	<b>Term Plan</b>	Term Plan	<b>Term Plan</b>
\$600	\$0.0680	\$0.0670	\$0.0660
\$2,400	\$0.0680	\$0.0670	\$0.0660
\$6,000	\$0.0680	\$0.0670	\$0.0660
\$9,000	\$0.0680	\$0.0670	\$0.0660
\$12,000	\$0.0680	\$0.0670	\$0.0660
\$18,000	\$0.0680	\$0.0670	\$0.0660
\$24,000	\$0.0680	\$0.0670	\$0.0660
\$30,000	\$0.0680	\$0.0670	\$0.0660
\$42,000	\$0.0680	\$0.0670	\$0.0660
\$60,000	\$0.0680	\$0.0670	\$0.0640
\$90,000	\$0.0680	\$0.0670	\$0.0640
\$120,000	\$0.0680	\$0.0660	\$0.0620
\$180,000	\$0.0680	\$0.0650	\$0.0610
\$240,000	\$0.0660	\$0.0640	\$0.0600

SBC Long Distance, LLC	Attachment I
d/b/a AT&T Long Distance	Original Page 32
NY PSC Tariff No. 1 – Telephone	

- 1.7 Custom Business Services (continued)
  - 1.7.27 AT&T High Volume Calling II Plus<sup>1</sup> (continued)

<sup>1</sup> This Service is no longer available for new Customer term plan agreements effective June 27, 2005. Existing Customers may add, move, remove or change lines and/or locations for the duration of their current term plan agreement. This Service is no longer available to existing Customers upon expiration of the Customer's Term Plan.

- (A) Outbound Calls (continued)
  - .2 AT&T High Volume Dedicated Outbound Calling II Plus

2 Year 1 Year 3 Year MAC Term Plan Term Plan **Term Plan** \$600 \$0.0470 \$0.0460 \$0.0450 \$2,400 \$0.0460 \$0.0450 \$0.0440 \$6.000 \$0.0450 \$0.0440 \$0.0430 \$9,000 \$0.0450 \$0.0440 \$0.0430 \$12,000 \$0.0440 \$0.0430 \$0.0420 \$0.0440 \$18,000 \$0.0430 \$0.0420 \$24,000 \$0.0440 \$0.0430 \$0.0420 \$30,000 \$0.0430 \$0.0420 \$0.0410 \$42,000 \$0.0430 \$0.0420 \$0.0410 \$60,000 \$0.0420 \$0.0410 \$0.0400 \$90,000 \$0.0420 \$0.0410 \$0.0400 \$120,000 \$0.0410 \$0.0400 \$0.0390 \$180,000 \$0.0390 \$0.0400 \$0.0380 \$240.000 \$0.0390 \$0.0380 \$0.0370

The per minute usage rates are as follows.

SBC Long Distance, LLC	Attachment I
d/b/a AT&T Long Distance	Original Page 33
NY PSC Tariff No. 1 – Telephone	

- 1.7 Custom Business Services (continued)
  - 1.7.27 AT&T High Volume Calling II Plus<sup>1</sup> (continued)

<sup>1</sup> This Service is no longer available for new Customer term plan agreements effective June 27, 2005. Existing Customers may add, move, remove or change lines and/or locations for the duration of their current term plan agreement. This Service is no longer available to existing Customers upon expiration of the Customer's Term Plan.

- (B) Inbound Toll Free Calls
  - .1 AT&T High Volume Toll Free Calling II Plus Usage Rates
    - .a Without CMR

The per minute usage rates are the same as Section 1.7.27 (A).1 of the Price List.

.b With CMR

The rate is \$.016 per minute which applies in addition to the per minute usage rates in Section 1.7.27 (A).1 of the Price List.

- .2 AT&T High Volume Dedicated Toll Free Calling II Plus Usage Rates
  - .a Without CMR

The per minute usage rates are the same as Section 1.7.27 (A).2 of the Price List.

.b With CMR

The rate is \$.016 per minute which applies in addition to the per minute usage rates in Section 1.7.27 (A).2 of this Price List.

.3 Optional Feature Charges

The description and rates for available optional features may be found in the Company's interstate Voice Product Reference and Pricing Guide which may be found at <u>www.att.com/guidebook</u>.

SBC Long Distance, LLC	
d/b/a AT&T Long Distance	
NY PSC Tariff No. 1 – Telephone	Su

Attachment I 3<sup>rd</sup> Revised Page 34 Superseding 2<sup>nd</sup> Revised Page 34

### PRICE LIST

#### 1.7 Custom Business Services (continued)

1.7.27 AT&T High Volume Calling II Plus<sup>1</sup> (continued)

- (C) Out of Term Rates
  - .1 Switched/Dedicated Inter/intraLATA

Out Of Term per minute usage rates for Switched and Dedicated usage associated with 1-Year, 2-Year and 3-Year Term Plans are as follows:

Per Minute Rate	Switched	Dedicated
Month to Month	Out of Term	Out of Term
\$600	\$0.1105(I)	\$0.0763(I)
\$2400	\$0.1105(I)	\$0.0748(I)
\$6,000	\$0.1105(I)	\$0.0731(I)
\$9,000	\$0.1271(I)	\$0.0841(I)
\$ 12,000	\$0.1271(I)	\$0.0821(I)
\$ 18,000	\$0.1271(I)	\$0.0821(I)
\$ 24,000	\$0.1271(I)	\$0.0821(I)
\$ 30,000	\$0.1271(I)	\$0.0803(I)
\$ 42,000	\$0.1271(I)	\$0.0803(I)
\$ 60,000	\$0.1271(I)	\$0.0784(I)
\$ 90,000	\$0.1271(I)	\$0.0784(I)
\$120,000	\$0.1271(I)	\$0.0766(I)
\$180,000	\$0.1271(I)	\$0.0746(I)

<sup>1</sup>This Service is no longer available for new Customer term plan agreements effective June 27, 2005. Existing Customers may add, move, remove or change lines and/or locations for the duration of their current term plan agreement. This Service is no longer available to existing Customers upon expiration of the Customer's Term Plan.

Issued: January 4, 2010

Donna Daniele, Area Manager 208 South Akard Street, Dallas, Texas 75202 Effective: February 12, 2010

SBC Long Distance, LLC
d/b/a AT&T Long Distance
NY PSC Tariff No. 1 – Telephone

Effective: April 27, 2009

#### PRICE LIST

- 1.7 Custom Business Services (continued)
  - 1.7.28 Reserved for future use.
  - 1.7.29 Business Domestic Saver 15 Plus 1 Year<sup>1</sup>

<sup>1</sup>This service is no longer available to new Customers or existing Customers at new locations effective April 1, 2004.

The usage rate for outbound calls and TFS calls is \$0.0800 per minute. The usage rate for fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, category 11 is \$0.15 per minute. The per call charge may be found in Section 1.1.1 (B).2.a, Section 1.1.2 (B), and Section 1.1.2 (C) of this Price List.

1.7.30 Business Long Distance 50 Plus 1 Year<sup>1</sup>

<sup>1</sup>This service is no longer available to new Customers or existing Customers at new locations effective April 1, 2004.

The usage rate for outbound calls and TFS calls is \$0.0800 per minute. The usage rate for fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, category 11 is \$0.15 per minute. The per call charge may be found in Section 1.1.1 (B).2.a, Section 1.1.2 (B), and Section 1.1.2 (C) of this Price List.

1.7.31 Business Long Distance 100 Plus 1 Year<sup>1</sup>

<sup>1</sup>This service is no longer available to new Customers or existing Customers at new locations effective April 1, 2004.

The usage rate for outbound calls and TFS calls is \$0.0800 per minute. The usage rate for fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, category 11 is \$0.14 per minute. The per call charge may be found in Section 1.1.1 (B).2.a, Section 1.1.2 (B), and Section 1.1.2 (C) of this Price List.

1.7.32 Business Domestic Saver 15 Plus 2 Year<sup>1</sup>

<sup>1</sup>This service is no longer available to new Customers or existing Customers at new locations effective April 1, 2004.

The usage rate for outbound calls and TFS calls is \$0.0800 per minute. The usage rate for fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, category 11 is \$0.15 per minute. The per call charge may be found in Section 1.1.1 (B).2.a, Section 1.1.2 (B), and Section 1.1.2 (C) of this Price List.

1.7.33 Business Long Distance 50 Plus 2 Year<sup>1</sup>

<sup>1</sup>This service is no longer available to new Customers or existing Customers at new locations effective April 1, 2004.

The usage rate for outbound calls and TFS calls is \$0.0800 per minute. The usage rate for fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, category 11 is \$0.15 per minute. The per call charge may be found in Section 1.1.1 (B).2.a, Section 1.1.2 (B), and Section 1.1.2 (C) of this Price List.

SBC Long Distance, LLC AT&T Long Distance d/b/a AT&T Long Distance NY PSC Tariff No. 5 – Telephone Attachment I Original Page 36

Effective: April 27, 2009

#### PRICE LIST

- 1.7 Custom Business Services (continued)
  - 1.7.34 Business Long Distance 100 Plus 2 Year<sup>1</sup>

<sup>1</sup>This service is no longer available to new Customers or existing Customers at new locations effective April 1, 2004.

The usage rate for outbound calls and TFS calls is \$0.0800 per minute. The usage rate for fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, category 11 is \$0.14 per minute. The per call charge may be found in Section 1.1.1 (B).2.a, Section 1.1.2 (B), and Section 1.1.2 (C) of this Price List.

1.7.35 Business Domestic Saver 15 Connections 1 Plus Service 1 Year<sup>1</sup>

<sup>1</sup>This service is no longer available to new Customers or existing Customers at new locations effective April 1, 2004.

The usage rate for outbound calls and TFS calls is \$0.0680 per minute. The usage rate for fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, category 11 is \$0.15 per minute. The per call charge may be found in Section 1.1.1 (B).2.a, Section 1.1.2 (B), and Section 1.1.2 (C) of this Price List.

1.7.36 Business Long Distance 50 Connections 1 Plus Service 1 Year<sup>1</sup>

<sup>1</sup>This service is no longer available to new Customers or existing Customers at new locations effective April 1, 2004.

The usage rate for outbound calls and TFS calls is \$0.0680 per minute. The usage rate for fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, category 11 is \$0.15 per minute. The per call charge may be found in Section 1.1.1 (B).2.a, Section 1.1.2 (B), and Section 1.1.2 (C) of this Price List.

1.7.37 Business Long Distance 100 Connections 1 Plus Service 1 Year<sup>1</sup>

<sup>1</sup>This service is no longer available to new Customers or existing Customers at new locations effective April 1, 2004.

The usage rate for outbound calls and TFS calls is \$0.0680 per minute. The usage rate for fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, category 11 is \$0.14 per minute. The per call charge may be found in Section 1.1.1 (B).2.a, Section 1.1.2 (B), and Section 1.1.2 (C) of this Price List.

SBC Long Distance, LLC
d/b/a AT&T Long Distance
NY PSC Tariff No. 1 – Telephone

#### 1.7 Custom Business Services (continued)

1.7.38 Business Domestic Saver 15 Connections 1 Plus Service 2 Year<sup>1</sup>

The usage rate for outbound calls and TFS calls is \$0.0680 per minute. The usage rate for fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, category 11 is \$0.15 per minute. The per call charge may be found in Section 1.1.1 (B).2.a, Section 1.1.2 (B), and Section 1.1.2 (C) of this Price List.

1.7.39Business Long Distance 50 Connections 1 Plus Service 2 Year<sup>1</sup>

The usage rate for outbound calls and TFS calls is \$0.0680 per minute. The usage rate for fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, category 11 is \$0.15 per minute. The per call charge may be found in Section 1.1.1 (B).2.a, Section 1.1.2 (B), and Section 1.1.2 (C) of this Price List.

1.7.40 Business Long Distance 100 Connections 1 Plus Service 2 Year<sup>1</sup>

The usage rate for outbound calls and TFS calls is \$0.0680 per minute. The usage rate for fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, category 11 is \$0.14 per minute. The per call charge may be found in Section 1.1.1 (B).2.a, Section 1.1.2 (B), and Section 1.1.2 (C) of this Price List.

1.7.41 Business Domestic Saver 15 Connections 2 Plus Service 1 Year<sup>1</sup>

The usage rate for outbound calls and TFS calls is \$0.0680 per minute. The usage rate for fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, category 11 is \$0.15 per minute. The per call charge may be found in Section 1.1.1 (B).2.a, Section 1.1.2 (B), and Section 1.1.2 (C) of this Price List.

1.7.42 Business Long Distance 50 Connections 2 Plus Service 1 Year<sup>1</sup>

The usage rate for outbound calls and TFS calls is \$0.0680 per minute. The usage rate for fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, category 11 is \$0.15 per minute. The per call charge may be found in Section 1.1.1 (B).2.a, Section 1.1.2 (B), and Section 1.1.2 (C) of this Price List.

1.7.43 Business Long Distance 100 Connections 2 Plus Service 1 Year<sup>1</sup>

The usage rate for outbound calls and TFS calls is \$0.0680 per minute. The usage rate for fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, category 11 is \$0.14 per minute. The per call charge may be found in Section 1.1.1 (B).2.a, Section 1.1.2 (B), and Section 1.1.2 (C) of this Price List.

<sup>1</sup>This service is no longer available to new Customers or existing Customers at new locations effective April 1, 2004.

SBC Long Distance, LLC	Attachment I
d/b/a AT&T Long Distance	Original Page 38
NY PSC Tariff No. 1 – Telephone	

- 1.7 Custom Business Services (continued)
  - 1.7.44 Business Domestic Saver 15 Connections 2 Plus Service 2 Year<sup>1</sup>

The usage rate for outbound calls and TFS calls is \$0.0680 per minute. The usage rate for fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, category 11 is \$0.15 per minute. The per call charge may be found in Section 1.1.1 (B).2.a, Section 1.1.2 (B), and Section 1.1.2 (C) of this Price List.

1.7.45 Business Long Distance 50 Connections 2 Plus Service 2 Year<sup>1</sup>

The usage rate for outbound calls and TFS calls is \$0.0680 per minute. The usage rate for fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, category 11 is \$0.15 per minute. The per call charge may be found in Section 1.1.1 (B).2.a, Section 1.1.2 (B), and Section 1.1.2 (C) of this Price List.

1.7.46 Business Long Distance 100 Connections 2 Plus Service 2 Year<sup>1</sup>

The usage rate for outbound calls and TFS calls is \$0.0680 per minute. The usage rate for fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, category 11 is \$0.14 per minute. The per call charge may be found in Section 1.1.1 (B).2.a, Section 1.1.2 (B), and Section 1.1.2 (C) of this Price List.

<sup>1</sup>This service is no longer available to new Customers or existing Customers at new locations effective April 1, 2004.

SBC Long Distance, LLC	Attachment I
d/b/a AT&T Long Distance	2 <sup>nd</sup> Revised Page 39
NY PSC Tariff No. 1 – Telephone	Superseding 1 <sup>st</sup> Revised Page 39

- 1.7 Custom Business Services (continued)
  - 1.7.47 Value Plans<sup>1</sup>
    - (A) AT&T Business Calling Value \$15 formerly Business Domestic Value Saver 15

The per minute usage rates for outbound and TFS calls are as follows:

<b>Customer Commitment</b>	<b>Rate Per Minute</b>
1 Year Term Plan	\$0.0800
2 Year Term Plan	\$0.0800

For fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, category 11 the usage rate is \$0.15 per minute. The per call charge may be found in Section 1.1.1 (B).2.a, Section 1.1.2 (B), and Section 1.1.2 (C) of this Price List.

(B) AT&T Business Calling Value \$50 formerly Business Long Distance Value 50

The per minute usage rates for outbound and TFS calls are as follows:

<b>Customer Commitment</b>	<b>Rate Per Minute</b>
1 Year Term Plan	\$0.0800
2 Year Term Plan	\$0.0800

For fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, category 11 the usage rate is \$0.15 per minute. The per call charge may be found in Section 1.1.1 (B).2.a, Section 1.1.2 (B), and Section 1.1.2 (C) of this Price List.

(C) AT&T Business Calling Value \$100 formerly Business Long Distance Value 100

The per minute usage rates for outbound and TFS calls are as follows:

<b>Customer Commitment</b>	<b>Rate Per Minute</b>
1 Year Term Plan	\$0.0800
2 Year Term Plan	\$0.0800

For fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, category 11 the usage rate is \$0.14 per minute. The per call charge may be found in Section 1.1.1 (B).2.a, Section 1.1.2 (B), and Section 1.1.2 (C) of this Price List.

<sup>1</sup>This Service is no longer available for new or to existing Customers or existing Customers at new locations effective November 2, 2009.

Т

SBC Long Distance, LLC
d/b/a AT&T Long Distance
NY PSC Tariff No. 1 – Telephone

Attachment I Original Page 40

Effective: April 27, 2009

## PRICE LIST

- 1.7 Custom Business Services (continued)
  - 1.7.48 AT&T Business Unlimited Calling Plans formerly Business Unlimited Long Distance Plans

AT&T Business Unlimited Calling Plans

The per minute usage rate for switched TFS is as follows:

Switched TFS Rate Per Minute \$0.0900

For fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, the usage rate is \$0.1400 per minute. The per call charge may be found in Section 1.1.1 (B).2.a, Section 1.1.2 (B), and Section 1.1.2 (C) of this Price List.

The MRC for unlimited interstate and intrastate 1+ outbound calling is as follows:

Number of Access Line Subscribed to AT&T Business Unlimited Calling Plans	MRC
1	\$20
2	\$40
3	\$60
4	\$80
5	\$100
6	\$120
7	\$140
8	\$160
9	\$180
10	\$200

1.7.49 Business Long Distance Solutions<sup>1</sup>

<sup>1</sup> This service no longer available to new Customers or existing Customers at new locations effective June 12, 2005. This product is also not available to existing Customers after the expiration of the term plan agreement.

The per minute usage rates for outbound and switched TFS calls and for fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2 category 11 are listed in the table below. The per call charges may be found in Section 1.1.1 (B).2.a, Section 1.1.2 (B), and Section 1.1.2 (C) of this Price List.

Rate Option	1-Year Term	2-Year Term
Business Long Distance Solutions 15		
Outbound and Switched TFS	\$0.0800	\$0.0800
Calling Card - Option 2, Category 11	\$0.1500	\$0.1500
Business Long Distance Solutions 50		
Outbound and Switched TFS	\$0.0800	\$0.0800
Calling Card - Option 2, Category 11	\$0.1500	\$0.1500
Business Long Distance Solutions 100		
Outbound and Switched TFS	\$0.0800	\$0.0800
Calling Card - Option 2, Category 11	\$0.1400	\$0.1400

SBC Long Distance, LLC	Attachment I
d/b/a AT&T Long Distance	2 <sup>nd</sup> Revised Page 41
NY PSC Tariff No. 1 – Telephone	Superseding 1 <sup>st</sup> Revised Page 41

- 1.7 Custom Business Services (continued)
  - 1.7.50 Reserved for future use
  - 1.7.51 AT&T Business Block of Time formerly Signature Block of Time

The MRCs for each block of time Business Optional Calling Plan are shown in the table below in the column labeled MRC. The per minute rates for intrastate Direct-Dialed outbound one-plus (1+), Toll Free Service, and fully-automated, operator assisted, and operator dialed calls billed to the Calling Card B Option 2 Category 12 after the block of time has been used is shown in the table below in the column labeled Rate Over Block.

AT&T Business Block of Time Rate Plan	MRC	<b>Rate Over Block</b>
1000 MOUs 1-Year Term <sup>1</sup>	\$39	\$0.052
1000 MOUs 2-Year Term <sup>1</sup>	\$39	\$0.050
1000 MOUs 3-Year Term <sup>1</sup>	\$39	\$0.048
2500 MOUs 1-Year Term <sup>1</sup>	\$90	\$0.048
2500 MOUs 2-Year Term <sup>1</sup>	\$90	\$0.046
2500 MOUs 3-Year Term <sup>1</sup>	\$90	\$0.044
5000 MOUs 1-Year Term <sup>1</sup>	\$175	\$0.046
5000 MOUs 2-Year Term <sup>1</sup>	\$175	\$0.044
5000 MOUs 3-Year Term <sup>1</sup>	\$175	\$0.042
7500 MOUs 1-Year Term <sup>2</sup> (C)	\$255	\$0.044
7500 MOUs 2-Year Term <sup>2</sup> (C)	\$255	\$0.042
7500 MOUs 3-Year Term <sup>2</sup> (C)	\$255	\$0.040
10000 MOUs 1-Year Term <sup>2</sup> (C)	\$320	\$0.042
10000 MOUs 2-Year Term <sup>2</sup> (C)	\$320	\$0.040
10000 MOUs 3-Year Term <sup>2</sup> (C)	\$320	\$0.038

The per call charge for operator assisted and operator dialed calling cards billed to the Calling Card B Option 2, Category 12 may be found in Section 1.1.1 (B).2.a, Section 1.1.2 (B), and Section 1.1.2 (C) of this Price List.

<sup>1</sup>This block of time is no longer available for new Customer term plan agreements or to existing Customers who N make changes to their existing service or move to a new location effective September 12, 2009. N

<sup>2</sup>This block of time is no longer available for new Customer term plan agreements or to existing Customers who N make changes to their existing service or move to a new location effective April 1, 2010. N

Issued: February 1, 2010

SBC Long Distance, LLC		tance, LLC	Attachment I	
d/b/a /	AT&T Lo	ong Distance	1 <sup>st</sup> Revised Page 42	
NY PS	SC Tariff	No. 1 – Telephone	Superseding Original Page 42	
		PRICE LIST		
1.7	Custon	n Business Services (continued)		
	1.7.52	AT&T Business Calling \$5.95 1-Year formerly	Business Domestic Saver 1-Year	С
		The per minute usage rates for outbound and sy automated, operator assisted and operator diale 2, Category 11 are listed below. The per call ch (B).2.a, Section 1.1.2 (B), and Section 1.1.2 (C	d calls billed to the Calling Card - Option arges may be found in Section 1.1.1	
		AT&T Business Calling \$5.95 1-Year		
		Outbound and Switched TFS Calling Card - Option 2, Category 11	\$0.0800 \$0.1500	
1.7.53		AT&T Business Calling \$15 Advantage former Year	rly Business Domestic Saver Solutions 1-	C
		The per minute usage rates for outbound and sy operator assisted, and operator dialed calls bille 11 are listed below. The per call charges may b 1.1.2 (B), and Section 1.1.2 (C) of this Price Li	ed to the Calling Card - Option 2, Category e found in Section 1.1.1 (B).2.a, Section	
		AT&T Business Calling \$15 Advantage		
		Outbound and Switched TFS Calling Card - Option 2, Category 11	\$0.0900 \$0.1500	

<sup>1</sup>This Service is no longer available to new Customers or existing Customers at new locations effective N May 12, 2009. N

Attachment I
1 <sup>st</sup> Revised Page 43
Superseding Original Page 43

#### 1.7 Custom Business Services (continued)

SBC Long Distance, LLC

d/b/a AT&T Long Distance

NY PSC Tariff No. 1 - Telephone

## 1.7.54 AT&T High Volume Calling III<sup>1, 2</sup>

(A) Outbound Calls

.2

.1 AT&T High Volume Outbound Calling III

The per minute usage rates are as follows:

MAC	1 Year	2 Year	3 Year
MAC	Term Plan	Term Plan	Term Plan
\$600	\$0.0680	\$0.0670	\$0.0660
\$2,400	\$0.0680	\$0.0670	\$0.0660
\$6,000	\$0.0680	\$0.0670	\$0.0660
\$9,000	\$0.0680	\$0.0670	\$0.0660
\$12,000	\$0.0680	\$0.0670	\$0.0660
\$18,000	\$0.0680	\$0.0670	\$0.0660
\$24,000	\$0.0680	\$0.0670	\$0.0660
\$30,000	\$0.0680	\$0.0670	\$0.0660
\$42,000	\$0.0680	\$0.0670	\$0.0660
\$60,000	\$0.0680	\$0.0670	\$0.0640
\$90,000	\$0.0680	\$0.0670	\$0.0640
\$120,000	\$0.0680	\$0.0660	\$0.0620
\$180,000	\$0.0680	\$0.0650	\$0.0610
\$240,000	\$0.0660	\$0.0640	\$0.0600

AT&T High	Volume Dedicated	Outbound Calling III

The per minute usage rates are as follows:

MAG	1 Year	2 Year	3 Year
MAC	Term Plan	Term Plan	Term Plan
\$600	\$0.0470	\$0.0460	\$0.0450
\$2,400	\$0.0460	\$0.0450	\$0.0440
\$6,000	\$0.0450	\$0.0440	\$0.0430
\$9,000	\$0.0450	\$0.0440	\$0.0430
\$12,000	\$0.0440	\$0.0430	\$0.0420
\$18,000	\$0.0440	\$0.0430	\$0.0420
\$24,000	\$0.0440	\$0.0430	\$0.0420
\$30,000	\$0.0430	\$0.0420	\$0.0410
\$42,000	\$0.0430	\$0.0420	\$0.0410
\$60,000	\$0.0420	\$0.0410	\$0.0400
\$90,000	\$0.0420	\$0.0410	\$0.0400
\$120,000	\$0.0410	\$0.0400	\$0.0390
\$180,000	\$0.0400	\$0.0390	\$0.0380
\$240,000	\$0.0390	\$0.0380	\$0.0370

<sup>1</sup>Effective November 12, 2007, the Dedicated Service Offering associated with High Volume Calling will no longer be available to new Customers.

<sup>2</sup>This Service is no longer available for new Customer term plan agreements effective July 12, 2009. Existing Customers may add, move, remove or change lines and/or locations for the duration of their current term plan agreement. This Service is no longer available to existing Customers upon the expiration of their current term plan agreement.

Effective: July 12, 2009

С

SBC Long Distance, LLC	Attachment I
d/b/a AT&T Long Distance	1 <sup>st</sup> Revised Page 44
NY PSC Tariff No. 1 – Telephone	Superseding Original Page 43

### PRICE LIST

### 1.7 Custom Business Services (continued)

- 1.7.54 AT&T High Volume Calling III<sup>1, 2</sup> (continued)
  - (B) Inbound Toll Free Calls
    - .1 AT&T High Volume Toll Free Calling III- Usage Rates
      - .a Without CMR

The per minute usage rates are the same as Section 1.7.54 (A).1 of this Price List.

.b With CMR

The rate is \$.016 per minute which applies in addition to the per minute usage rates in Section 1.7.54 (A).1 of this Price List.

- .2 AT&T High Volume Dedicated Toll Free Calling III Usage Rates
  - .a Without CMR

The per minute usage rates are the same as Section 1.7.54 (A).2 of this Price List.

.b With CMR

The rate is \$.016 per minute which applies in addition to the per minute usage rates in Section 1.7.54 (A).2 of this Price List.

.3 Optional Feature Charges

The description and rates for available optional features may be found in the Company's interstate Voice Product Reference and Pricing Guide which may be found at <u>www.att.com</u>.

<sup>1</sup>Effective November 12, 2007, the Dedicated Service Offering associated with High Volume Calling will no longer be available to new Customers.

<sup>2</sup>This Service is no longer available for new Customer term plan agreements effective July 12, 2009. Existing Customers may add, move, remove or change lines and/or locations for the duration of their current term plan agreement. This Service is no longer available to existing Customers upon the expiration of their current term plan agreement.

Issued: May 27, 2009

SBC Long Distance, LLC	Attachment I
d/b/a AT&T Long Distance	1 <sup>st</sup> Revised Page 44.1
NY PSC Tariff No. 1 – Telephone	Superseding Original Page 44.1

### 1.7 Custom Business Services (continued)

1.7.54 AT&T High Volume Calling III<sup>1, 2</sup> (continued)

(C) Out of Term (OOT)

OOT per minute usage rates for intrastate interLATA and intrastate intraLATA calls associated with 1, 2, and 3 year term agreements are as

MAC	PER MINUTE RATE		
MAC	SWITCHED	DEDICATED	
\$600	\$0.1001(I)	\$0.0687(I)	
\$2,400	\$0.1001(I)	\$0.0672(I)	
\$6,000	\$0.1001(I)	\$0.0658(I)	
\$9,000	\$0.1001(I)	\$0.0658(I)	
\$12,000	\$0.1001(I)	\$0.0643(I)	
\$18,000	\$0.1001(I)	\$0.0643(I)	
\$24,000	\$0.1001(I)	\$0.0643(I)	
\$30,000	\$0.1001(I)	\$0.0628(I)	
\$42,000	\$0.1001(I)	\$0.0628(I)	
\$60,000	\$0.1001(I)	\$0.0613(I)	
\$90,000	\$0.1001(I)	\$0.0613(I)	
\$120,000	\$0.1001(I)	\$0.0599(I)	
\$180,000	\$0.1001(I)	\$0.0584(I)	

<sup>1</sup>Effective November 12, 2007, the Dedicated Service Offering associated with High Volume Calling will no longer be available to new Customers.

<sup>2</sup>This Service is no longer available for new Customer term plan agreements effective July 12, 2009. Existing Customers may add, move, remove or change lines and/or locations for the duration of their current term plan agreement. This Service is no longer available to existing Customers upon the expiration of their current term plan agreement.

Issued: January 4, 2010

Effective: February 12, 2010

Effective: April 27, 2009

SBC Long Distance, LLC	Attachment l
d/b/a AT&T Long Distance	Original Page 45
NY PSC Tariff No. 1 – Telephone	

### PRICE LIST

### 1.7 Custom Business Services (continued)

1.7.55 AT&T Business Calling \$15 Advantage formerly Business Domestic Saver 15 Prime<sup>SM</sup>

The per minute usage rate for domestic outbound 1+ and switched TFS calls and for fully automated, operator assisted, and operator dialed calls billed to the calling Card B Option 2, Category 11 are listed in the table below. The per call charges may be found in Section 1.1.1 (B).2.a, Section 1.1.2 (B), and Section 1.1.2 (C) of this Price List.

Rate Options	1-Year Term	2-Year Term
Outbound 1+ & Switched TFS	\$0.0800	\$0.0800
Calling Card B Option 2, Category 11	\$0.1500	\$0.1500

- 1.7.56 Business Unlimited Prime Long Distance Plans<sup>1</sup>
  - (A) For fully automated, operator assisted, and operator dialed calls billed to the Calling Card
     Option 2, the usage rate is \$0.14 per minute. The per call charges may be found in Section 1.1.1 (B).2.a, Section 1.1.2 (B), and Section 1.1.2 (C) of this Price List.
- 1.7.57 AT&T Long Distance Aggregation Preferred<sup>2</sup>
  - (A) Switched/Dedicated Inter/intraLATA Rates

	Per Minute Rate		
MAC	1-Year	2-Year	3-Year
	<b>Term Plan</b>	Term Plan	Term Plan
\$ 9,000	\$0.0744	\$0.0744	\$0.0744
\$ 12,000	\$0.0744	\$0.0744	\$0.0744
\$ 18,000	\$0.0744	\$0.0744	\$0.0744
\$ 24,000	\$0.0744	\$0.0744	\$0.0744
\$ 30,000	\$0.0744	\$0.0744	\$0.0744
\$ 42,000	\$0.0744	\$0.0744	\$0.0744
\$ 60,000	\$0.0744	\$0.0744	\$0.0744
\$ 90,000	\$0.0744	\$0.0744	\$0.0744
\$120,000	\$0.0744	\$0.0744	\$0.0744
\$180,000	\$0.0744	\$0.0744	\$0.0744

<sup>1</sup>This Service is no longer available to new Customers or existing Customers at new locations effective February 12, 2007

<sup>2</sup>This Service is no longer available to new Customers or existing Customers at new locations effective June 1, 2008.

SBC Long Distance, LLC
d/b/a AT&T Long Distance
NY PSC Tariff No. 1 – Telephone

Attachment I Original Page 46

## PRICE LIST

- 1.7 Custom Business Services (continued)
  - 1.7.58 AT&T Long Distance Aggregation Preferred II
    - (A) Outbound Calls
      - .1 The per minute usage rates are as follows

MAC	1 Year Rate	2 Year Rate	3 Year Rate
\$ 9,000	\$0.0432	\$0.0402	\$0.0384
\$ 12,000	\$0.0405	\$0.0387	\$0.0375
\$ 18,000	\$0.0404	\$0.0386	\$0.0374
\$ 24,000	\$0.0403	\$0.0385	\$0.0373
\$ 30,000	\$0.0402	\$0.0384	\$0.0372
\$ 42,000	\$0.0385	\$0.0374	\$0.0356
\$ 60,000	\$0.0384	\$0.0373	\$0.0355
\$ 90,000	\$0.0378	\$0.0372	\$0.0354
\$ 120,000	\$0.0373	\$0.0356	\$0.0343
\$ 180,000	\$0.0372	\$0.0355	\$0.0342

- (B) Inbound Toll Free Calls
  - .1 The per minute usage rates are the same as Section 4.7.54 (A).1 of this Tariff.
  - .2 Optional Feature Charges For optional feature charges, see Section 4.6.2 of this Tariff.
- (C) Calls Billed To Calling Card Option 3

The per-minute usage rates are the same as Section 4.7.54 (A).1 of this Tariff.

SBC Long Distance, LLC d/b/a AT&T Long Distance NY PSC Tariff No. 1 – Telephone Attachment I 1<sup>st</sup> Revised Page 46.1 Superseding Original Page 46.1

## PRICE LIST

- 1.7 Custom Business Services (continued)
  - 1.7.59 AT&T Business Block of Time<sup>SM</sup> 200 II

Calling Card Option 2 calls are billed in increments of one (1) second subject to a minimum connect time of thirty (30) seconds. The usage rate is \$0.18 per minute and the per call charges are as follows:

- Fully Automated \$1.25
- Operator Assisted \$1.95
- Operator Dialed \$2.95
- 1.7.60 AT&T Business Calling <sup>SM</sup> Monthly
  - (A) The bundled outbound intrastate/interstate and/or inbound toll free calling per minute usage rate is \$0.12 until March 12, 2010 at which time the per minute T usage rate will increase to \$0.14.
  - (B) The bundled outbound intrastate/interstate MRC is \$8.00. See the Company's Voice Product Reference and Pricing Guide at <u>http://att.com/servicepublications</u> T Section 1.6.4 for the Switched Toll Free Service MRC.
  - (C) For fully automated, operator assisted, and operator dialed calls billed to Calling Card - Option 2, the usage rate is \$0.20 per minute and the per call charges are as follows:
    - Fully Automated \$1.25
    - Operator Assisted \$1.95
    - Operator Dialed \$2.95

Issued: January 15, 2010

SBC Long Distance, LLC	Attachment I
d/b/a AT&T Long Distance	2 <sup>nd</sup> Revised Page 46.2
NY PSC Tariff No. 1 – Telephone	Superseding 1 <sup>st</sup> Revised Page 46.2

### 1.7 Custom Business Services (continued)

1.7.61 AT&T High Volume Calling IV

The per minute usage rates for outbound calls and inbound TFS calls are as follows:

	InterLATA		IntraLATA			
MAC	1 Year	2 Year	3 Year	1 Year	2 Year	3 Year
MAC	Term Plan					
\$600	\$0.0680	\$0.0670	\$0.0660	\$0.	\$0.	\$0.
\$2,400	\$0.0680	\$0.0670	\$0.0660	\$0.	\$0.	\$0.
\$6,000	\$0.0680	\$0.0670	\$0.0660	\$0.	\$0.	\$0.
\$9,000	\$0.0680	\$0.0670	\$0.0660	\$0.	\$0.	\$0.
\$12,000	\$0.0680	\$0.0670	\$0.0660	\$0.	\$0.	\$0.
\$18,000	\$0.0680	\$0.0670	\$0.0660	\$0.	\$0.	\$0.
\$24,000	\$0.0680	\$0.0670	\$0.0660	\$0.	\$0.	\$0.
\$30,000	\$0.0680	\$0.0670	\$0.0660	\$0.	\$0.	\$0.
\$42,000	\$0.0680	\$0.0670	\$0.0660	\$0.	\$0.	\$0.
\$60,000	\$0.0680	\$0.0670	\$0.0640	\$0.	\$0.	\$0.
\$90,000	\$0.0680	\$0.0670	\$0.0640	\$0.	\$0.	\$0.
\$120,000	\$0.0680	\$0.0660	\$0.0620	\$0.	\$0.	\$0.
\$180,000	\$0.0680	\$0.0650	\$0.0610	\$0.	\$0.	\$0.
\$240,000	\$0.0660	\$0.0640	\$0.0600	\$0.	\$0.	\$0.

1.7.62 AT&T Business Unlimited Calling<sup>SM</sup>II

Calling Card Option - 2 calls are billed in increments of six (6) seconds subject to a minimum connect time of thirty (30) seconds. For fully automated, operator assisted, and operator dialed calls billed to Calling Card - Option 2, the usage rate is \$0.18 per minute and the per call charges are as follows:

- Fully Automated \$1.25
- Operator Assisted \$1.95
- Operator Dialed \$2.95

Issued: June 23, 2009

T T

Т

Ν

Effective: September 1, 2009

SBC Long Distance, LLC d/b/a AT&T Long Distance <u>NY PSC Tariff No. 1 – Telephone</u> Attachment I Original Page 46.3

### PRICE LIST

- 1.7 Custom Business Services (continued)
  - 1.7.63 Block of Time Term Agreement Plans

### **Rate Options**

- (A) AT&T Business Block of Time<sup>SM</sup> 700 II
  - .1 1-Year Term Agreement

Calling Card Option - 2 calls are billed in increments of one (1) second subject to a minimum connect time of thirty (30) seconds. For fully automated, operator assisted, and operator dialed calls billed to Calling Card - Option 2, the usage rate is \$0.18 per minute and the per call charges are as follows:

- Fully Automated \$1.25
- Operator Assisted \$1.95
- Operator Dialed \$2.95
- .2 2-Year Term Agreement

Calling Card Option - 2 calls are billed in increments of one (1) second subject to a minimum connect time of thirty (30) seconds. For fully automated, operator assisted, and operator dialed calls billed to Calling Card - Option 2, the usage rate is \$0.18 per minute and the per call charges are as follows:

- Fully Automated \$1.25
- Operator Assisted \$1.95
- Operator Dialed \$2.95

Issued: July 21, 2009

Ν

Effective: September 1, 2009

## PRICE LIST

- 1.7 Custom Business Services (continued)
  - 1.7.63 Block of Time Term Agreement Plans (continued)

#### **Rate Options (continued)**

- (B) AT&T Business Block of Time<sup>SM</sup> 1200 II
  - .1 1-Year Term Agreement

Calling Card Option - 2 calls are billed in increments of one (1) second subject to a minimum connect time of thirty (30) seconds. For fully automated, operator assisted, and operator dialed calls billed to Calling Card - Option 2, the usage rate is \$0.18 per minute and the per call charges are as follows:

- Fully Automated \$1.25
- Operator Assisted \$1.95
- Operator Dialed \$2.95
- .2 2-Year Term Agreement

Calling Card Option - 2 calls are billed in increments of one (1) second subject to a minimum connect time of thirty (30) seconds. For fully automated, operator assisted, and operator dialed calls billed to Calling Card - Option 2, the usage rate is \$0.18 per minute and the per call charges are as follows:

- Fully Automated \$1.25
- Operator Assisted \$1.95
- Operator Dialed \$2.95

Issued: July 21, 2009

Ν

Effective: September 1, 2009

## PRICE LIST

- 1.7 Custom Business Services (continued)
  - 1.7.63 Block of Time Term Agreement Plans (continued)

#### **Rate Options (continued)**

- (C) AT&T Business Block of Time<sup>SM</sup> 2500 II
  - .1 1-Year Term Agreement

Calling Card Option - 2 calls are billed in increments of one (1) second subject to a minimum connect time of thirty (30) seconds. For fully automated, operator assisted, and operator dialed calls billed to Calling Card - Option 2, the usage rate is \$0.18 per minute and the per call charges are as follows:

- Fully Automated \$1.25
- Operator Assisted \$1.95
- Operator Dialed \$2.95
- .2 2-Year Term Agreement

Calling Card Option - 2 calls are billed in increments of one (1) second subject to a minimum connect time of thirty (30) seconds. For fully automated, operator assisted, and operator dialed calls billed to Calling Card - Option 2, the usage rate is \$0.18 per minute and the per call charges are as follows:

- Fully Automated \$1.25
- Operator Assisted \$1.95
- Operator Dialed \$2.95
- .3 3-Year Term Agreement

Calling Card Option - 2 calls are billed in increments of one (1) second subject to a minimum connect time of thirty (30) seconds. For fully automated, operator assisted, and operator dialed calls billed to Calling Card - Option 2, the usage rate is \$0.18 per minute and the per call charges are as follows:

- Fully Automated \$1.25
- Operator Assisted \$1.95
- Operator Dialed \$2.95

SBC Long Distance, LLC
d/b/a AT&T Long Distance
NY PSC Tariff No. 1 – Telephone

Attachment I Original Page 46.6

## PRICE LIST

- 1.7 Custom Business Services (continued)
  - 1.7.63 Block of Time Term Agreement Plans (continued)

#### **Rate Options (continued)**

- (D) AT&T Business Block of Time<sup>SM</sup> 5000 II
  - .1 1-Year Term Agreement

Calling Card Option - 2 calls are billed in increments of one (1) second subject to a minimum connect time of thirty (30) seconds. For fully automated, operator assisted, and operator dialed calls billed to Calling Card - Option 2, the usage rate is \$0.18 per minute and the per call charges are as follows:

- Fully Automated \$1.25
- Operator Assisted \$1.95
- Operator Dialed \$2.95
- .2 2-Year Term Agreement

Calling Card Option - 2 calls are billed in increments of one (1) second subject to a minimum connect time of thirty (30) seconds. For fully automated, operator assisted, and operator dialed calls billed to Calling Card - Option 2, the usage rate is \$0.18 per minute and the per call charges are as follows:

- Fully Automated \$1.25
- Operator Assisted \$1.95
- Operator Dialed \$2.95
- .3 3-Year Term Agreement

Calling Card Option - 2 calls are billed in increments of one (1) second subject to a minimum connect time of thirty (30) seconds. For fully automated, operator assisted, and operator dialed calls billed to Calling Card - Option 2, the usage rate is \$0.18 per minute and the per call charges are as follows:

- Fully Automated \$1.25
- Operator Assisted \$1.95
- Operator Dialed \$2.95

Ν

Ν

SBC Long Distance, LLC
d/b/a AT&T Long Distance
NY PSC Tariff No. 1 - Telephone

Attachment I Original Page 46.7

### PRICE LIST

1.7 Custom Business Services (continued)

1.7.64 AT&T Business Unlimited Calling<sup>SM</sup>III

Calling Card Option - 2 calls as defined in Section 3.1.5 of this Tariff are billed in increments of six (6) seconds subject to a minimum connect time of thirty (30) seconds. For fully automated, operator assisted, and operator dialed calls billed to Calling Card - Option 2, the usage rate is \$0.18 per minute and the per call charges are as follows:

- Fully Automated \$1.25
- Operator Assisted \$1.95
- Operator Dialed \$2.95

Issued: September 17, 2009

С

N N

## SBC Long Distance, LLC d/b/a AT&T Long Distance <u>NY PSC Tariff No. 1 – Telephone</u>

## PRICE LIST

### 1.8 Custom Consumer Services

- 1.8.1 Reserved for future use
- 1.8.2 Reserved for future use
- 1.8.3 Reserved for future use
- 1.8.4 Reserved for future use
- 1.8.5 Reserved for future use
- 1.8.6 Reserved for future use
- 1.8.7 Reserved for future use
- 1.8.8 AT&T Nationwide Calling 120<sup>SM</sup> Direct<sup>1</sup>

The bundled intrastate/interstate MRC is \$11.99 The per minute usage rate is \$0.10 per minute for calls completed after the 120 minute block of time has been used.

1.8.9 AT&T ONE RATE<sup>®</sup> Nationwide10 Cents Direct

The bundled interstate/intrastate MRC is \$2.99 and the per minute usage rate is \$0.10.

<sup>1</sup>This Service is no longer available to new Customers or existing Customers at new locations effective July 12, 2009.

SBC Long Distance, LLC
d/b/a AT&T Long Distance
NY PSC Tariff No. 1 – Telephone

#### 1.9 Grandfathered Services

1.9.1 Business Long Distance<sup>1</sup>

<sup>1</sup>This Service is no longer available to new Customers or to existing Customers at new locations effective February 1, 2001.

The per minute usage rate for InterLATA and IntraLATA calls calls is \$0.53.

For fully automated, operator assisted, and operator dialed calls billed to the Calling Card - Option 2, the usage rate is \$0.53 per minute. The per call charge may be found in Section 4.1.1 (B).2.a, Section 4.1.2 (C) of this Tariff.

1.9.2 Business Long Distance Total Solutions<sup>1</sup>

<sup>1</sup>This Service is no longer available to new Customers or to existing Customers at new locations effective February 1, 2001.

The usage rate is \$0.10 per minute.

- 1.9 Grandfathered Services (continued)
  - 1.9.3 High Volume Calling Connections I<sup>1</sup>

<sup>1</sup>This service is no longer available to new Customers or existing Customers at new locations effective April 1, 2005.

- (A) Outbound Calls
  - .1 High Volume Outbound Calling Connections I

The per minute usage rates for intrastate InterLATA calls are as follows:

	Per Minute Rate		
MAC	1-Year	2-Year	3-Year
	Term Plan	Term Plan	Term Plan
\$600	\$0.0820	\$0.0780	\$0.0740
\$2,400	\$0.0800	\$0.0760	\$0.0720
\$6,000	\$0.0780	\$0.0740	\$0.0700
\$12,000	\$0.0760	\$0.0720	\$0.0680
\$30,000	\$0.0740	\$0.0700	\$0.0660
\$60,000	\$0.0720	\$0.0680	\$0.0640
\$120,000	\$0.0700	\$0.0660	\$0.0620
\$180,000	\$0.0680	\$0.0650	\$0.0610
\$240,000	\$0.0660	\$0.0640	\$0.0600

The per minute usage rates for intrastate IntraLATA calls are as follows:

	Per Minute Rate		
MAC	1-Year	2-Year	3-Year
	<b>Term Plan</b>	<b>Term Plan</b>	<b>Term Plan</b>
\$600	\$0.0820	\$0.0780	\$0.0740
\$2,400	\$0.0800	\$0.0760	\$0.0720
\$6,000	\$0.0780	\$0.0740	\$0.0700
\$12,000	\$0.0760	\$0.0720	\$0.0680
\$30,000	\$0.0740	\$0.0700	\$0.0660
\$60,000	\$0.0720	\$0.0680	\$0.0640
\$120,000	\$0.0700	\$0.0660	\$0.0620
\$180,000	\$0.0680	\$0.0650	\$0.0610
\$240,000	\$0.0660	\$0.0640	\$0.0600

Effective: April 27, 2009

## PRICE LIST

- 1.9 Grandfathered Services (continued)
  - 1.9.3 High Volume Calling Connections  $I^1$  (continued)

<sup>1</sup>This service is no longer available to new Customers or existing Customers at new locations effective April 1, 2005.

- (A) Outbound Calls (continued)
  - .2 High Volume Dedicated Outbound Calling Connections I

The per minute usage rates for InterLATA calls are as follows.

	Per Minute Rate		
MAC	1-Year	2-Year	3-Year
	<b>Term Plan</b>	<b>Term Plan</b>	<b>Term Plan</b>
\$600	\$0.0640	\$0.0600	\$0.0560
\$2,400	\$0.0620	\$0.0580	\$0.0540
\$6,000	\$0.0600	\$0.0560	\$0.0520
\$12,000	\$0.0580	\$0.0540	\$0.0500
\$30,000	\$0.0560	\$0.0520	\$0.0480
\$60,000	\$0.0540	\$0.0500	\$0.0460
\$120,000	\$0.0520	\$0.0480	\$0.0440
\$180,000	\$0.0500	\$0.0470	\$0.0430
\$240,000	\$0.0480	\$0.0460	\$0.0420

The per minute usage rates for IntraLATA calls are as follows.

	Per Minute Rate		
MAC	1-Year	2-Year	3-Year
	<b>Term Plan</b>	<b>Term Plan</b>	<b>Term Plan</b>
\$600	\$0.0640	\$0.0600	\$0.0560
\$2,400	\$0.0620	\$0.0580	\$0.0540
\$6,000	\$0.0600	\$0.0560	\$0.0520
\$12,000	\$0.0580	\$0.0540	\$0.0500
\$30,000	\$0.0560	\$0.0520	\$0.0480
\$60,000	\$0.0540	\$0.0500	\$0.0460
\$120,000	\$0.0520	\$0.0480	\$0.0440
\$180,000	\$0.0500	\$0.0470	\$0.0430
\$240,000	\$0.0480	\$0.0460	\$0.0420

SBC Long Distance, LLC
d/b/a AT&T Long Distance
NY PSC Tariff No. 1 - Telephone

- 1.9 Grandfathered Services (continued)
  - 1.9.3 High Volume Calling Connections  $I^1$  (continued)

<sup>1</sup>This service is no longer available to new Customers or existing Customers at new locations effective April 1, 2005.

- (B) Inbound Toll Free Calls
  - .1 High Volume Toll Free Calling Connections I Usage Rates
    - .a Without CMR

The per minute usage rates are the same as Section 1.9.3 (A).1 of this Tariff.

.b With CMR

The rate is \$.03 per minute which applies in addition to the per minute usage rates in Section 1.9.3 (A).1 of this Tariff.

- .2 High Volume Dedicated Toll Free Calling Connections I Usage Rates
  - .a Without CMR

The per minute usage rates are the same as Section 1.7.3 (A).2 of this Tariff.

.b With CMR

The rate is 0.3 per minute which applies in addition to the per minute usage rates in Section 1.7.3 (A).2 of this Tariff.

.3 Optional Feature Charges

The description and rates for available optional features may be found in the Company's interstate Voice Product Reference and Pricing Guide which may be found at <u>www.att.com/guidebook</u>.

- 1.9 Grandfathered Services (continued)
  - 1.9.4 High Volume Calling Connections II<sup>1</sup>

<sup>1</sup>This service is no longer available to new Customers or existing Customers at new locations effective April 1, 2005.

- (A) Outbound Calls
  - .1 High Volume Outbound Calling Connections II

The per minute usage rates for intrastate InterLATA calls are as follows:

	Per Minute Rate		
MAC	1-Year	2-Year	3-Year
	<b>Term Plan</b>	<b>Term Plan</b>	<b>Term Plan</b>
\$600	\$0.0820	\$0.0780	\$0.0740
\$2,400	\$0.0800	\$0.0760	\$0.0720
\$6,000	\$0.0780	\$0.0740	\$0.0700
\$12,000	\$0.0760	\$0.0720	\$0.0680
\$30,000	\$0.0740	\$0.0700	\$0.0660
\$60,000	\$0.0720	\$0.0680	\$0.0640
\$120,000	\$0.0700	\$0.0660	\$0.0620
\$180,000	\$0.0680	\$0.0650	\$0.0610
\$240,000	\$0.0660	\$0.0640	\$0.0600

The per minute usage rates for intrastate IntraLATA calls are as follows:

	Per Minute Rate		
MAC	1-Year	2-Year	3-Year
	<b>Term Plan</b>	<b>Term Plan</b>	<b>Term Plan</b>
\$600	\$0.0820	\$0.0780	\$0.0740
\$2,400	\$0.0800	\$0.0760	\$0.0720
\$6,000	\$0.0780	\$0.0740	\$0.0700
\$12,000	\$0.0760	\$0.0720	\$0.0680
\$30,000	\$0.0740	\$0.0700	\$0.0660
\$60,000	\$0.0720	\$0.0680	\$0.0640
\$120,000	\$0.0700	\$0.0660	\$0.0620
\$180,000	\$0.0680	\$0.0650	\$0.0610
\$240,000	\$0.0660	\$0.0640	\$0.0600

- 1.9 Grandfathered Services (continued)
  - 1.9.4 High Volume Calling Connections II<sup>1</sup> (continued)

<sup>1</sup>This service is no longer available to new Customers or existing Customers at new locations effective April 1, 2005.

- (A) Outbound Calls (continued)
  - .2 High Volume Dedicated Outbound Calling Connections II

The per minute usage rates for InterLATA calls are as follows.

	Per Minute Rate		
MAC	1-Year	2-Year	3-Year
	<b>Term Plan</b>	<b>Term Plan</b>	<b>Term Plan</b>
\$600	\$0.0640	\$0.0600	\$0.0560
\$2,400	\$0.0620	\$0.0580	\$0.0540
\$6,000	\$0.0600	\$0.0560	\$0.0520
\$12,000	\$0.0580	\$0.0540	\$0.0500
\$30,000	\$0.0560	\$0.0520	\$0.0480
\$60,000	\$0.0540	\$0.0500	\$0.0460
\$120,000	\$0.0520	\$0.0480	\$0.0440
\$180,000	\$0.0500	\$0.0470	\$0.0430
\$240,000	\$0.0480	\$0.0460	\$0.0420

The per minute usage rates for IntraLATA calls are as follows.

	Per Minute Rate		
MAC	1-Year	2-Year	3-Year
	<b>Term Plan</b>	<b>Term Plan</b>	<b>Term Plan</b>
\$600	\$0.0640	\$0.0600	\$0.0560
\$2,400	\$0.0620	\$0.0580	\$0.0540
\$6,000	\$0.0600	\$0.0560	\$0.0520
\$12,000	\$0.0580	\$0.0540	\$0.0500
\$30,000	\$0.0560	\$0.0520	\$0.0480
\$60,000	\$0.0540	\$0.0500	\$0.0460
\$120,000	\$0.0520	\$0.0480	\$0.0440
\$180,000	\$0.0500	\$0.0470	\$0.0430
\$240,000	\$0.0480	\$0.0460	\$0.0420

SBC Long Distance, LLC
d/b/a AT&T Long Distance
NY PSC Tariff No. 1 – Telephone

Attachment I Original Page 54

Effective: April 27, 2009

### PRICE LIST

# 1.9 Grandfathered Services (continued)

1.9.4 High Volume Calling Connections II<sup>1</sup> (continued)

<sup>1</sup>This service is no longer available to new Customers or existing Customers at new locations effective April 1, 2005.

- (B) Inbound Toll Free Calls
  - .1 High Volume Toll Free Calling Connections II Usage Rates
    - .a Without CMR

The per minute usage rates are the same as Section 1.9.4 (A).1 of this Tariff.

.b With CMR

The rate is \$.03 per minute which applies in addition to the per minute usage rates in Section 1.9.4 (A).1 of this Tariff.

- .2 High Volume Dedicated Toll Free Calling Connections II Usage Rates
  - .a Without CMR

The per minute usage rates are the same as Section 1.9.4 (A).2 of this Tariff.

.b With CMR

The rate is \$.03 per minute which applies in addition to the per minute usage rates in Section 1.9.4 (A).2 of this Tariff.

.3 Optional Feature Charges

The description and rates for available optional features may be found in the Company's interstate Voice Product Reference and Pricing Guide which may be found at <u>www.att.com/guidebook</u>.

SBC Long Distance, LLC
d/b/a AT&T Long Distance
NY PSC Tariff No. 1 – Telephone

Attachment I 2<sup>nd</sup> Revised Page 55 Superseding 1<sup>st</sup> Revised Page 55

## PRICE LIST

### 1.10 Miscellaneous Charges

1.10.1 Return Check Charge

The rate is \$20 per occurrence.

- 1.10.2 Reserved for future use.
- 1.10.3 Reserved for future use.
- 1.10.4 Payphone Origination Charge

The rate is \$0.60 per call.

Gross Revenue Tax Surcharge\*

Period	Surcharge
January 1, 1997 - September 30, 1997	3.6269%
October 1, 1998 - December 31, 1999	3.3592%
January 1, 2000 +	2.5641%

MTA Tax Surcharge\*

 Period
 Surcharge

 1997 +
 .5986%

\*These surcharges are not applicable to Services provided for resale to telecommunications companies possessing a Certificate of Public Convenience and Necessity from the New York State Public Service Commission, or designated as eligible for a sale-for-resale exclusion from the New York State Department of Taxation and Finance.

1.10.5 Account Codes

	MRC
Small Business Plans	\$10.00 per BTN/BAN
High Volume Calling Plans	\$00.00 per BTN/BAN

Issued: February 1, 2010

Т

Т

Attachment I Original Page 56

Effective: April 27, 2009

# PRICE LIST

## A. SERVICES INELIGIBLE FOR SCHOOLS AND LIBRARIES DISCOUNT

1. Voice Mail Services

## B. SCHOOLS AND LIBRARIES DISCOUNT MATRIX

% of students eligible for national school lunch program

HOW	%DISCOUNT LEVEL	
DISADVANTAGED	Urban Discount	<b>Rural Discount</b>
<1	20	25
1-19	40	50
20-34	50	60
35-49	60	70
50-74	80	80
75-100	90	90